### SEITEL CANADA HOLDINGS, INC.

- and -

**PULSE SEISMIC INC.** 

#### **SHARE PURCHASE AGREEMENT**

**DATED JANUARY 15, 2019** 

### TABLE OF CONTENTS

Page

# ARTICLE 1 PURCHASE OF SHARES

1.1	Purchase and Sale				
1.2	Closing				
1.3	Deliveries of the Vendor	1			
1.4	Deliveries of the Purchaser				
1.5	Amount of Purchase Price				
1.6	Estimated Purchase Price				
1.7	Preparation of Closing Balance Sheet and Determination of Closing Working	∠			
1.7	Capital	2			
1.8	Acceptance or Dispute of the Closing Balance Sheet and the Closing Working				
1.0	Capital	3			
1.9	Post-Closing Adjustments	4			
1.10	Payment of Purchase Price				
1.11	A/R Cash Adjustment				
1.12	Cash Resale Adjustment				
	ARTICLE 2 REPRESENTATIONS AND WARRANTIES				
	REFRESENTATIONS AND WARRANTIES				
2.1	Representations and Warranties of the Vendor	7			
2.2	Representations and Warranties of the Purchaser	17			
2.3	Commissions				
2.4	No Additional Representations or Warranties				
	ARTICLE 2				
	ARTICLE 3 SURVIVAL AND INDEMNIFICATION				
3.1	Survival	20			
3.2	Indemnity by the Vendor				
3.3	Indemnity by the Purchaser				
3.4	Claim Notice				
3.5	Monetary Limitations				
3.6	Certain Additional Matters				
3.7	Direct Claims				
3.8	Third Party Claims				
3.9	Manner of Payment				
3.10	Exclusive Remedy	24			
	ARTICLE 4				
	COVENANTS				
4.1	Preparation of Tax Returns	25			
4.2	Audit of the Corporation				
4.3	Director and Officer Liability				
4.4	Access to Books and Records				
4.5	Conflicts				
٦.٥	Oo iiiioto	21			

# TABLE OF CONTENTS (continued)

		Page
4.6	Use of Name	28
4.7	Tax Refunds	
4.8	Releases.	
4.9	Bonuses; Severance	
4.10	Corporation Liabilities	
	ARTICLE 5 GENERAL	
	GENERAL	
5.1	Actions on Non-Business Days	30
5.2	Currency and Payment Obligations	30
5.3	Calculation of Time	30
5.4	Schedules and Exhibits	30
5.5	Expenses	31
5.6	Public Announcements	
5.7	Notices	31
5.8	Time of Essence	32
5.9	Further Assurances	32
5.10	Entire Agreement	
5.11	Amendment	33
5.12	Waiver	33
5.13	Severability	33
5.14	Governing Law; Jurisdiction	
5.15	Successors and Assigns; Assignment	34
5.16	Third Party Beneficiaries	34
5.17	Counterparts	34
5.18	Specific Performance	34
5.19	Non-Recourse	34
5.20	Non-Competition	35

# SCHEDULE 1.0 DEFINITIONS

SCHEDULE 2.0 ADDITIONAL RULES OF INTERPRETATION.

#### SHARE PURCHASE AGREEMENT

This Share Purchase Agreement dated January 15, 2019 is made

#### BETWEEN

Seitel Canada Holdings, Inc. (the "Vendor")

- and -

Pulse Seismic Inc. (the "Purchaser")

#### **RECITALS**

- A. The Vendor is the registered and beneficial owner of the Shares;
- B. The Purchaser is willing to purchase and the Vendor is willing to sell the Shares on and subject to the terms and conditions contained in this Agreement;
- C. Certain defined terms used in, and other clauses pertaining to the interpretation of, this Agreement are set out in Schedule 1.0;
- D. Simultaneously with the execution of this Agreement, the Vendor and/or Affiliate(s) of the Vendor and the Corporation are executing the transition services agreement attached hereto as <u>Exhibit A</u> (the "**Transition Services Agreement**"); and
- E. Simultaneously with the execution of this Agreement, the Purchaser, the Vendor and the Escrow Agent are executing the escrow agreement attached hereto as <a href="Exhibit B">Exhibit B</a> (the "Escrow Agreement"), pursuant to which an escrow account (the "Escrow Account") shall have been established pursuant to the terms and conditions thereof.

For good and valuable consideration, the receipt and adequacy of which are hereby acknowledged by each Party, the Parties agree as follows.

### Article 1 PURCHASE OF SHARES

- **1.1 Purchase and Sale**. At the Closing Time, on and subject to the terms and conditions of this Agreement, the Vendor hereby sells to the Purchaser, and the Purchaser hereby purchases from the Vendor, the Shares.
- **1.2 Closing**. The Closing shall take place at 10:00 AM Eastern Time on the date hereof (the "**Closing Time**") by telephone conference and electronic exchange of documents (or, if the Parties mutually agree in writing to hold a physical closing, at the offices of Kirkland & Ellis LLP, located at 300 North LaSalle Street, Chicago, Illinois 60654), or on such other date and at such other time as may be mutually agreed upon by the Parties in writing.
- **1.3 Deliveries of the Vendor**. At the Closing, the Vendor shall deliver:
- (1) Shares. The Shares to the Purchaser, together with any certificates representing the Shares and any customary and duly executed instrument of transfer (e.g., "stock powers");

- (2) Books and Records. Possession of the Books and Records of the Corporation at the offices and facilities of the Corporation and access to such offices and facilities;
- (3) Resignations. The resignation of each officer and director of the Corporation listed on Schedule 1.3(3); and
- (4) Data. A copy of the Seismic Data delivered in the format in which it is currently stored, using quality redundant storage appliances as a means of transfer.

#### **1.4 Deliveries of the Purchaser**. At the Closing, the Purchaser shall deliver:

- (1) Estimated Purchase Price and Escrow Amount. The Purchaser shall pay or direct to be paid the Estimated Purchase Price and the Escrow Amount in the manner provided in Section 1.10.
- **1.5 Amount of Purchase Price**. The aggregate price payable by the Purchaser to the Vendor for the Shares is \$53,600,000 (the "**Purchase Price**"), subject to adjustment as set forth herein.

#### 1.6 Estimated Purchase Price.

- (1) Estimated Working Capital. No later than three (3) Business Days prior to the Closing Date, the Vendor provided to the Purchaser an estimate of the Working Capital as at the open of business on the Closing Date, based on the Corporation's Books and Records and other information available at the Closing Date and calculated in a manner consistent with the manner in which the Closing Working Capital is to be determined under Section 1.7(1) (the "Estimated Working Capital").
  - (2) Expected Working Capital. The Purchase Price will be adjusted as follows:
  - (a) if the Estimated Working Capital is more than \$(415,000) (the "Expected Working Capital"), the Purchase Price will be increased, dollar for dollar, by the amount of the difference: or
  - (b) if the Estimated Working Capital is less than the Expected Working Capital, the Purchase Price will be decreased, dollar for dollar, by the amount of the difference,

and the resulting amount in either case will be the "Estimated Purchase Price".

## 1.7 Preparation of Closing Balance Sheet and Determination of Closing Working Capital.

- (1) Within sixty (60) days following the Closing Date, the Purchaser shall cause the Purchaser's Auditor to prepare, at the Purchaser's expense, and deliver to the Vendor:
  - (a) the Closing Balance Sheet of the Corporation as at the open of business on the Closing Date, prepared in accordance with generally accepted accounting principles, applying the same significant accounting policies and practices as applied to the Annual Financial Statements (the "Transaction Accounting Principles") (the "Closing Balance Sheet"); and

- (b) a calculation of Working Capital as at the open of business on the Closing Date, calculated in accordance with the Transaction Accounting Principles, including the use of the same line items and line item entries, set forth on and used in preparation of the Sample Calculation (the "Closing Working Capital").
- The Closing Working Capital (A) will be determined in accordance with the (2)definitions set forth in this Agreement and using the Transaction Accounting Principles, and (B) (1) will not include any changes in assets or liabilities as a result of purchase accounting adjustments or other changes arising from or resulting as a consequence of the transactions contemplated by this Agreement, (2) will be based on facts and circumstances as they exist as of the Closing and will exclude the effects of any act, decision, change in circumstances or event arising or occurring on or after the Closing and (3) will not include any reserve or accrual not reflected in the Sample Calculation. The parties agree that the purpose of preparing the Closing Balance Sheet and determining the Closing Working Capital is solely to accurately measure changes (if any) in the amounts of Working Capital from the Estimated Working Capital in order to determine the Post-Closing Adjustment Amount, and that such processes are not intended to permit the introduction of principles, policies, practices, procedures, methodologies, classifications, methods, conventions, assumptions, judgments or estimation techniques that are different from those used in the calculation of Estimated Working Capital (including any exclusions or deviations from GAAP and the methodology used by the Vendor with respect to accruals and reserves to prepare such estimates).
- (3) Following receipt of the Closing Balance Sheet and the calculation of the Closing Working Capital, the Vendor will have the Review Period to review them. During the Review Period, the Purchaser shall ensure that the Purchaser's Auditor gives access, upon every reasonable request, to the Vendor and its Representatives, to the Books and Records and all working papers of the Purchaser and the Purchaser's Auditor created in connection with the preparation of the Closing Balance Sheet and the calculation of the Closing Working Capital, to verify the accuracy, presentation and other matters relating to the preparation of the Closing Balance Sheet and the calculation of the Closing Working Capital, and to enable the Vendor to exercise its rights under Section 1.8.
- (4) The Purchaser and the Vendor shall each bear their own fees and expenses, including the fees and expenses of their respective Auditors, in preparing or reviewing, as the case may be, the Closing Balance Sheet and the calculation of the Closing Working Capital. If a dispute regarding the Closing Balance Sheet or the calculation of the Closing Working Capital is submitted for determination to the Independent Accountant under Section 1.8(2), the fees and expenses of the Parties and the Independent Accountant will be paid in accordance with Section 1.8(2).

## 1.8 Acceptance or Dispute of the Closing Balance Sheet and the Closing Working Capital

(1) Following receipt of the Closing Balance Sheet and calculation of the Closing Working Capital as contemplated by Section 1.7, if the Vendor objects in good faith to any item on the Closing Balance Sheet or the calculation of the Closing Working Capital, the Vendor shall so notify the Purchaser by delivering a notice to that effect (the "Notice of Objection") to the Purchaser prior to the end of the Review Period. The Notice of Objection must set out the reasons for the Vendor's objection, the amount in dispute and reasonable details of the calculation of that amount. If the Vendor does not object to any item on the Closing Balance Sheet or the calculation of the Closing Working Capital, the Vendor may so notify the Purchaser

by delivering a notice to that effect (the "Notice of Acceptance"). If the Vendor does not deliver a Notice of Objection or Notice of Acceptance to the Purchaser prior to the end of the Review Period, the Vendor will be deemed to have delivered a Notice of Acceptance to the Purchaser on the last day of the Review Period and the calculation of the Closing Working Capital will be deemed to be final and binding for the purposes of the adjustment referred to in Section 1.9 (for greater certainty, neither Party is required to pay or entitled to receive as a result the adjustments referred to in Section 1.9 by making a claim for Damages under Article 3).

- If the Vendor delivers a Notice of Objection in accordance with Section 1.8(1), the Parties shall work expeditiously and in good faith to resolve all of the items in dispute set out in the Notice of Objection within 30 days following delivery of the Notice of Objection. Any items in dispute that are not resolved by the end of that 30 day period will be submitted by the Parties for determination to an independent accounting firm mutually agreed to by the Parties, or, if the Parties are unable to agree on the independent accounting firm within a further 10 day period, to Ernst & Young, LLC, or if that accounting firm is unable to act, to PricewaterhouseCoopers LLP. In making its determination, the selected accounting firm (the "Independent Accountant") will act as expert and not as arbitrator, must only consider the items in dispute submitted to it, and must make reasonable efforts to determine the items in dispute within 30 days following the date of submission. The determination of the Independent Accountant will be final and binding upon the Parties and will not be subject to appeal, absent manifest error, and the Closing Working Capital as determined by the Independent Accountant will be final and binding for purposes of the adjustments referred to in Section 1.9 (for greater certainty, neither Party may appeal a determination made by the Independent Accountant or attempt to dispute, recover, duplicate or adjust any amount which that Party is required to pay or entitled to receive as a result of that determination or the adjustment referred to in Section 1.9 by making a claim for Damages under Article 3). The Purchaser and the Vendor shall each pay one half of the fees and expenses of the Independent Accountant, but shall each bear their own costs in presenting their respective cases to the Independent Accountant.
- (3) Notwithstanding Article 3, the procedures set out in this Section 1.8 for resolving disputes with respect to the Closing Balance Sheet and the calculation of the Closing Working Capital is the sole and exclusive method of resolving those disputes, absent manifest error. However, this Section 1.8 will not:
  - (a) prohibit either Party from commencing litigation to compel specific performance of this Section 1.8 or to enforce the determination of the Independent Accountant; or
  - (b) prohibit either Party from exercising its rights under Section 3.2 or 3.3, as the case may be, if the basis of the dispute could also entitle that Party to make a claim for Damages under Section 3.2(a) or Section 3.3(a), as the case may be, and
  - (c) that claim is not discovered until after the delivery of the Closing Balance sheet under Section 1.7(1), and
  - (d) that claim involves Fraud of the other Party.

#### 1.9 Post-Closing Adjustments.

(1) The Purchase Price will be adjusted as follows:

- (a) if the Closing Working Capital, as finally determined in accordance with Section 1.8, is more than the Estimated Working Capital, the Purchase Price will be increased, dollar for dollar, by the amount of the difference; or
- (b) if the Closing Working Capital, as finally determined in accordance with Section 1.8, is less than the Estimated Working Capital, the Purchase Price will be decreased, dollar for dollar, by the amount of the difference.

The amount by which the Closing Working Capital, as finally determined in accordance with Section 1.8, is more or less than the Estimated Working Capital, is referred to as the "Post-Closing Adjustment Amount".

- (2) If the Closing Working Capital, as finally determined in accordance with Section 1.8, is more than the Estimated Working Capital, the Purchaser shall pay the Post-Closing Adjustment Amount together with interest on that amount as provided in Section 1.9(3) to the Vendor as an increase in the Purchase Price. If the Closing Working Capital, as finally determined in accordance with Section 1.8, is less than the Estimated Working Capital, the Vendor shall pay the Post-Closing Adjustment Amount together with interest on that amount as provided in Section 1.9(3) to the Purchaser as a decrease in the Purchase Price. The Purchaser or the Vendor, as the case may be, shall pay the Post-Closing Adjustment Amount together with interest on that amount as provided in this Section 1.9 on the Adjustment Date by wire transfer of immediately available funds to or to the order of the payee.
- (3) The Post-Closing Adjustment Amount will be paid together with interest on that amount calculated from the Closing Date to the date of payment at the Prime Rate plus 2%.
- (4) The Parties agree to treat any payment to this Section 1.9 as an adjustment to the Purchase Price for all Tax purposes, except as otherwise required by Applicable Law.
- **1.10** Payment of Purchase Price. The Purchase Price will be paid and satisfied, subject to adjustment in accordance with this Article 1, Section 3.6(5) and Section 4.7, as follows:
- (1) at the Closing Time, the Purchaser shall pay an amount equal to the Estimated Purchase Price, minus the Escrow Amount, by wire transfer of immediately available funds to or to the order of the Vendor,
- (2) at the Closing Time, the Purchaser shall pay, by wire transfer of immediately available funds, in trust, an amount in cash equal to \$1,000,000 (the "Escrow Amount") to the Escrow Agent, to be deposited into the Escrow Account and held in escrow and distributed in accordance with the provisions of the Escrow Agreement; and
- (3) on the Adjustment Date, the Purchaser or the Vendor, as the case may be, shall pay the Post-Closing Adjustment Amount, if any, in accordance with Section 1.9.

#### 1.11 A/R Cash Adjustment.

(1) From and after the Closing, the Purchaser shall pay to Vendor, as an increase to the Purchase Price, (i) any and all amounts of cash that the Corporation receives ("A/R Cash") in respect of accounts receivable set forth on the A/R Schedule under the heading "Pre-LOI A/R" (the "Pre-LOI A/R" and the A/R Cash attributable thereto, the "Pre-LOI A/R Cash Amount") and (ii) a portion of A/R Cash in respect of accounts receivable set forth on the A/R

<u>Schedule</u> under the heading "Pre-Closing A/R" (the "**Pre-Closing A/R**" and the A/R Cash attributable thereto, the "**Pre-Closing A/R Cash Amount**") in an amount equal to (a) the amounts attributable to reproduction and delivery services performed by an Affiliate of Vendor as set forth on each invoice of the Corporation's customers ("**Service Fees**") plus (b) 50% of Pre-Closing A/R Cash Amount in excess of the Services Fees (such amount in addition to the Service Fees and Pre-LOI A/R Cash Amount, the "**A/R Cash Payment**").

From and after the Closing until all Pre-LOI A/R and Pre-Closing A/R has been collected, the Corporation shall (y) deliver to Vendor within five (5) Business Days after the end of each calendar month (each such month, an "A/R Period") a statement (each, an "A/R Cash Statement") setting forth in reasonable detail the Purchaser's calculation of the A/R Cash Payment for the applicable A/R Period, and (z) pay by wire transfer of immediately available funds to, the account(s) designated in writing by the Vendor the A/R Cash Payment for such A/R Period. Following the delivery of each A/R Cash Statement, and during any period of dispute thereafter with respect to any A/R Cash Statement, the Corporation and the Purchaser will cooperate reasonably with the Vendor and its Representatives in the review of such A/R Cash Statement (including by furnishing on a timely basis information in connection with such review) and provide the Vendor and its Representatives access to the books, records, supporting data, facilities and personnel of the Purchaser and the Corporation's accountants and other Representatives for purposes of their review of the A/R Cash Statements. The Purchaser agrees that following the Closing it will not, and it will cause the Corporation not to, take any actions with respect to the books, records, policies and procedures of the Corporation that would obstruct or prevent the preparation of the A/R Cash Statements or the calculation of the A/R Cash Payment, or the Vendor's review of the foregoing as provided in this Section 1.11. Following the Closing, Purchaser shall, use commercially reasonable efforts to collect, and shall not delay or postpone the collection of, the accounts that contribute to the Pre-Closing A/R Cash Amount or agree or negotiate with any party to extend the payment date of any such account or discount any such account.

#### 1.12 Cash Resale Adjustment.

- (1) From and after the Closing the Purchaser shall pay to the Vendor, as an increase to the Purchase Price, fifty percent (50%) of amounts attributable to Cash Resales ("Cash Resale Amount") entered into during the calendar years 2019 (but after the Closing Date) and 2020 (the "Cash Resale Term"); provided, that the aggregate amount paid to the Vendor pursuant to this Section 1.12 shall not exceed \$5,000,000.
- (2) From and after the Closing and until the end of the Cash Resale Term, the Corporation shall (y) deliver to the Vendor within two (2) Business Days after the end of each calendar quarter for the calendar years 2019 and 2020 (each such quarter, a "Cash Resale Period") a statement (each, a "Cash Resale Statement") setting forth in reasonable detail the Cash Resales for such Cash Resale Period and Purchaser's calculation of the Cash Resale Amount for the applicable Cash Resale Period, and (z) within five (5) Business Days of timely delivery of each Cash Resale Statement, pay by wire transfer of immediately available funds to, the account(s) designated in writing by the Vendor the Cash Resale Amount for such Cash Resale Period. Following the delivery of each Cash Resale Statement, and during any period of dispute thereafter with respect to any Cash Resale Statement, the Corporation and the Purchaser will cooperate reasonably with the Vendor and its Representatives in the review of such Cash Resale Statement (including by furnishing on a timely basis information in connection with such review) and provide the Vendor and its Representatives access to the books, records, supporting data, facilities and personnel of the Purchaser and the Corporation's

accountants and other Representatives for purposes of their review of the Cash Resale Statements. The Purchaser agrees that following the Closing it will not, and it will cause the Corporation not to, take any actions with respect to the books, records, policies and procedures of the Corporation that would obstruct or prevent the preparation of the Cash Resale Statements or the calculation of the Cash Resale Amount, or the Vendor's review of the foregoing as provided in this Section 1.12. Following the Closing, (i) Purchaser shall not, and shall cause the Corporation not to, take any action with the intent of preventing any portion of the Cash Resale Amounts from being earned or paid hereunder and (ii) during the Cash Resale Term, Purchaser shall use commercially reasonable efforts to pursue Cash Resales and continue to operate the business in the ordinary course, including the pursuit of Cash Resales, consistent with past practice.

If the Vendor disagrees with any part of the Purchaser's calculation of the Cash Resale Amount for any Cash Resale Period, the Vendor will notify the Purchaser in writing of such disagreement by setting forth the Vendor's calculation of the Cash Resale Amount for such Cash Resale Period (a "Cash Resale Objection Notice"). If a Cash Resale Objection Notice is not delivered to the Purchaser, then the Cash Resale Statement will be final, conclusive and binding upon, and non-appealable, by the Parties. During the fifteen (15) days immediately following the delivery of a Cash Resale Objection Notice, or such longer period as the Vendor and the Purchaser may agree in writing, the Vendor and the Purchaser will seek in good faith to resolve in writing any differences that they may have with respect to any matter specified in the Cash Resale Objection Notice. In the event that the Purchaser and the Vendor resolve in writing all such disagreements, the amount for the Cash Resale Payment so agreed in writing by the Purchaser and the Vendor will be final, conclusive and binding upon, and nonappealable, by the Parties. In the event that the Purchaser and the Vendor are unable to resolve all such disagreements within such fifteen (15) day period after the Purchaser's receipt of such Cash Resale Objection Notice, the Purchaser and the Vendor will submit such remaining disagreements to the Independent Accountant to make a determination with respect thereto in accordance with the dispute resolution procedures set forth in Section 1.8(2) as if they applied to resolution of such disagreements, mutatis mutandis.

## Article 2 REPRESENTATIONS AND WARRANTIES

- **2.1** Representations and Warranties of the Vendor. The Vendor represents and warrants to the Purchaser, except as set forth in the Schedules, as of the date hereof as follows.
- (1) Incorporation and Corporate Power of Vendor. The Vendor is a corporation duly incorporated, organized and existing under the laws of the State of Delaware. The Vendor has the corporate power, authority and capacity to execute and deliver this Agreement and all other agreements and instruments to be executed by it as contemplated herein and to perform its obligations hereunder and under all such other agreements and instruments.
- (2) Authorization and Enforceability. The execution, delivery and performance by the Vendor of this Agreement and all agreements and instruments to be executed and delivered by Vendor hereunder have been duly authorized by all necessary corporate action on the part of the Vendor and no other corporate actions on the part of the Vendor are necessary to authorize the execution, delivery or performance by the Vendor of this Agreement or such other agreements and instruments, and this Agreement, assuming due and valid authorization, execution and delivery hereof by the Purchaser, constitutes the valid and binding obligation of the Vendor enforceable against the Vendor in accordance with its terms, except as limited by

the application of bankruptcy, insolvency, reorganization, fraudulent conveyance, moratorium or other Applicable Laws relating to or affecting creditors' rights or general principles of equity.

- (3) Ownership of Shares. The Vendor is the registered and beneficial holder of 316,080,137 common shares of the Corporation (the "Shares"), being the only issued and outstanding shares of the Corporation, with good and marketable title thereto, free and clear of all Liens, charges, encumbrances and claims of any other Person. There are no restrictions of any kind on the transfer of the Shares except those set out in the articles of incorporation of the Corporation and applicable securities laws. The Shares have been validly issued in compliance with Applicable Law.
- (4) No Other Agreements or Options. Except for the Purchaser's rights in this Agreement, no other Person has any Contract or option or any right or privilege capable of becoming an agreement or option for:
  - (a) the purchase or other acquisition from the Vendor of any of the Shares;
  - (b) the purchase, subscription, allotment or issuance of any unissued shares or other securities of the Corporation; or
  - (c) other than in the ordinary course of business, the purchase or other acquisition from the Corporation of any of its undertaking, property or assets.
- (5) Organization and Operation of the Corporation. The Corporation is incorporated, organized and subsisting under the laws of the Province of Alberta. The Corporation does not have a direct or indirect Equity Interest in any other Person. The Vendor has made available to the Purchaser true and accurate copies of the organizational documents of the Corporation as amended to date.
- (6) Qualification to do Business. The Corporation is registered, licensed or otherwise qualified to do business under the laws of the Provinces of Alberta and British Columbia, being the only jurisdictions in which the location of the properties and assets owned by the Corporation or the nature of its business as now carried on requires registration, licensing or other qualification. The Corporation has all necessary corporate power, authority, and capacity to carry on its business as now carried on and to own or lease and operate its property and assets as now carried on and owned or leased and operated.
  - (7) Corporate Records. The Corporation has made available to the Purchaser:
  - (a) complete and accurate copies of the articles of the Corporation;
  - (b) complete and accurate copies of all by-laws of the Corporation, all of which were duly enacted and are in full force and effect;
  - (c) complete and accurate, in all material respects, copies of minutes of all meetings of the Corporation's shareholders, directors and committees of directors since the date of amalgamation;
  - (d) complete and accurate, in all material respects, copies of all resolutions in writing passed by the Corporation's shareholders, directors and committees of directors since the date of amalgamation; and

(e) the share certificate book, securities register, register of transfers, register of directors and register of officers of the Corporation, each of which is current and complete and accurate in all material respects,

each of which has been maintained in accordance with Applicable Law.

There are no shareholders' agreements or unanimous shareholders' agreements governing the affairs of the Corporation or the relationship, rights and duties of its shareholders or directors nor are there any voting trusts, pooling arrangements or other similar agreements with respect to the ownership or voting of any shares of the Corporation.

- (8) Investment Canada Act. The Corporation is not a cultural business as defined in the Investment Canada Act, or a business that falls within a specific type of business activity that, in the opinion of the Governor in Council, is related to Canada's cultural heritage or national identity as prescribed under the Investment Canada Act.
- (9) Competition Act (Canada). With respect to Part IX of the Competition Act, the aggregate value of the assets in Canada of the Corporation are less than \$92 million and the gross revenues from sales in or from Canada from such assets is less than \$92 million in aggregate value, all as determined in accordance with the Competition Act and the regulations thereunder.
- (10) Bankruptcy, Insolvency and Reorganization. The Corporation is not an insolvent person within the meaning of the Bankruptcy and Insolvency Act (Canada) nor has the Corporation made an assignment in favour of its creditors nor a proposal in bankruptcy to its creditors or any class thereof nor had any petition for a receiving Order presented in respect of it. The Corporation has not initiated proceedings with respect to a compromise or arrangement with its creditors or for its winding up, liquidation or dissolution. No receiver has been appointed in respect of the Corporation or any of its property or assets and no execution or distress has been levied upon any of its property or assets. No act or proceeding has been taken or authorized by or against the Corporation with respect to any amalgamation, merger, consolidation, arrangement or reorganization of, or relating to, the Corporation, nor have any such proceedings been authorized by the Corporation, the Vendor or any of their Affiliates.
  - (11) Financial Statements.
  - (a) Except as set forth on Schedule 2.1(11)(a), the Financial Statements have been prepared in accordance with GAAP consistently applied throughout the periods to which they relate, subject, in the case of the Interim Financial Statements, to usual year-end adjustments and the exclusion of footnotes. The balance sheets contained in the Financial Statements fairly present the financial position of the Corporation as of their respective dates and the statements of earnings and retained earnings contained in the Financial Statements fairly present the revenues, earnings and results of operations for the periods indicated.
  - (b) Except as set forth on Schedule 2.1(11)(b), the Corporation does not have any liabilities or obligations of the type required to be accrued on or reserved against in a consolidated balance sheet prepared in accordance with GAAP (collectively, "Liabilities"), except (i) Liabilities accrued on or reserved against in the Financial Statements or disclosed in the notes thereto or in the notes to the other Financial Statements, (ii) Liabilities that have arisen since the date of the latest Financial

Statement in the ordinary course of business, (iii) Liabilities to be included in the computation of Corporation Liabilities, (iv) Liabilities disclosed in another section of the Schedules, and (v) other Liabilities which would not reasonably be expect to have, individually or in the aggregate, a Material Adverse Change.

- (12) Absence of Developments. From the date of the Interim Financial Statements until the date hereof, other than as set forth in Schedule 2.1(12), there has not occurred any event, occurrence or development that has had, or would reasonably be expected to have, individually or in the aggregate, a Material Adverse Change.
  - (13) Seismic Data.
  - (a) Except as provided in Schedule 2.1(13)(a), the Corporation owns a 100% interest in the three dimensional Seismic Data and no third party has any ownership, revenue or other interest in the three dimensional Seismic Data.
  - (b) Except as provided in Schedule 2.1(13)(b), the Vendor has not licensed or otherwise given the Seismic Data to any third party, other than seismic data licenses entered into in the ordinary course of business and on normal industry terms (the "Seismic Data Licenses").
  - (c) The Seismic Data Licenses are in full force and effect, and there are no defaults by the Vendor or, to the best of knowledge of the Vendor, any licensee under the Seismic Data Licenses.
  - (d) Except as provided in Schedule 2.1(13)(d), the Seismic Data is not under any exclusivity period or other restriction on marketing or sale.
  - (e) The Seismic Data does not, to the best of the knowledge of the Vendor, infringe upon the intellectual property rights of any third party.
  - (f) All Joint Venture Agreements are in full force and effect, and there are no defaults by the Vendor or any other party to such joint venture and other agreements.
- (14) Title to Assets. The Corporation has good and marketable legal and beneficial title to all of its property and assets, free and clear of any and all Liens except for Permitted Liens and there is no agreement, option or other right or privilege outstanding in favour of any Person for the purchase from the Corporation of any of its property or assets other than the purchase of Seismic Data in the ordinary course of business. A complete list of all real property leased by the Corporation that requires aggregate annual payments in excess of \$5,000 per month is set forth on Schedule 2.1(14) (the "Leased Real Property"), and the Corporation does not own any real property. The Corporation has a valid leasehold estate in all of the Leased Real Property, free and clear of all Liens, other than Permitted Liens. The Corporation has not leased or otherwise granted to any Person rights to use or occupy any of the Leased Real Property that would reasonably be expected to materially impair the use or occupancy of the Leased Real Property in the operation of the business of the Corporation.
- (15) Compliance with Applicable Law. The Corporation is in compliance with Applicable Law.

- (16) Legal Proceedings and Orders. There is no Legal Proceeding in progress, pending or, to the knowledge of the Vendor or the Corporation, Threatened against or affecting the Corporation or any of its officers or directors in their capacity as such, or any of its property or assets or title thereto, which Legal Proceeding involves the possibility of any Damages not fully covered by insurance. There is no Order outstanding against or affecting the Corporation or any of its property or assets.
- (17) Contracts. Except as set forth on Schedule 2.1(17) as of the date hereof the Corporation is not a party to any:
  - (a) agreement for the employment of any officer, individual employee or other Person on a full-time or consulting basis;
  - agreement under which the Corporation has borrowed any money or issued any note, indenture or other evidence of indebtedness or guaranteed indebtedness or liabilities of others;
  - (c) agreement that prohibits the Corporation from engaging in business with respect to any geographic area or Person in a manner material to the Corporation;
  - (d) agreement relating to any acquisition or disposition by the Corporation of business (whether by asset or stock purchase or otherwise) or any merger, consolidation or similar business combination transaction:
  - (e) agreement for the delivery of goods or provision of services with an outstanding balance as of the date hereof payable to or by the Corporation; or
  - (f) agreement in writing to enter into any of the foregoing.
  - (18) Intellectual Property.
  - (a) Schedule 2.1(18) sets forth a correct and complete list of all Intellectual Property that is registered, filed or issued under the authority of any Governmental Authority, and all applications for Intellectual Property, in each case that is owned by the Corporation (collectively, "Corporation Intellectual Property"). Except as set forth on Schedule 2.1(18), the Corporation owns the Corporation Intellectual Property, free and clear of all Liens, other than Permitted Liens, except as is not material to the Corporation.
  - (b) To the knowledge of the Corporation, the Corporation's business as now carried on does not infringe, misappropriate or otherwise violate any Intellectual Property of any other Person, except where such infringement, misappropriation or violation is not material to the Corporation.
  - (c) To the knowledge of the Corporation, no third party infringes, misappropriates or otherwise violates any Intellectual Property owned by the Corporation.
  - (d) To the knowledge of the Corporation, all of the issued patents and registered trademarks that constitute Corporation Intellectual Property are valid, subsisting and enforceable. Except for office actions issued in the ordinary course of prosecution by the United States Patent and Trademark Office or analogous foreign Governmental Authority, during the prior two (2) years, no claim by any

third party contesting the validity or enforceability of any of the Corporation Intellectual Property has been made or has been Threatened, in each case in writing.

- (19) Environmental Matters.
- (a) The Corporation is, and has been during the prior two (2) years, in compliance in all material respects with all applicable Environmental Laws, which compliance has included obtaining and maintaining all permits, licenses and authorizations required under applicable Environmental Laws that are material to the operations of the Corporation as currently conducted.
- (b) To the knowledge of the Corporation, during the prior five (5) years no Hazardous Substance has been released at any property leased by the Corporation in violation of any Environmental Law, except for such release or violation that is not material to the Corporation taken as a whole.
- (c) The Corporation:
  - (i) has not been convicted of an offence, fined or otherwise sentenced for non-compliance with any Environmental Laws;
  - (ii) has not been investigated or subject to any Legal Proceeding for noncompliance with any Environmental Law;
  - (iii) is not and has not been subject to any Order or other sanction requiring investigation or remediation of any real/immovable property, or
  - (iv) has not settled any Legal Proceeding for non-compliance with any Environmental Law short of conviction in connection therewith.
- (d) The Corporation has not received any notice and, to the knowledge of the Vendor, there are no facts that could give rise to any notice, that the Corporation is potentially responsible for any remedial or other corrective action or any work, repairs, construction or capital expenditures to be made under any Environmental Law with respect to the Business, including any real/immoveable property previously owned, leased or used by the Corporation.
- (20) Employees and Employee Benefits.
- (a) Schedule 2.1(20)(a) contains a list of the titles or positions of all Employees with their date of hire and the location of their employment, whether they are actively at work or not and, if not, the reason for the absence and expected return to work date, a list of all written Contracts with Employees, and a complete and accurate list of the remuneration of, and Employee Plans applicable to, each Employee. Schedule 2.1(20)(a) also contains a list of all individuals receiving compensation for work or services provided to the Corporation who are not Employees and particulars of their terms of engagement.
- (b) Except as disclosed on Schedule 2.1(20)(b), to the knowledge of the Vendor, the Corporation has no Employee who cannot be dismissed on reasonable notice

and is not liable to any Employee or former employee for any damages under any Applicable Law or any agreement for arrangement relating to any employee benefits.

- (c) The Corporation has complied, in all material respects, with all Applicable Law and Orders applicable to it relating to employment of the Employees, including those relating to wages, hours of work, overtime and other employment standards, human rights, collective bargaining, occupational health and safety, workers' hazardous materials, pay equity and workers' compensation. Except as disclosed in Schedule 2.1(20)(c), there are no outstanding claims, complaints or proceedings by any Employee or former employees of the Corporation under any employment standards, human rights, pay equity, occupational health and safety, workplace safety and insurance or any other employment related statute, and the Corporation has not been advised that any such claims, complaints or proceedings may be filed.
- (d) The Corporation is not a party to or bound by, either directly or by operation of Applicable Law, any collective agreement, labour contract, letter of understanding, letter of intent, voluntary recognition agreement or legally binding commitment or written communication to any labour union, trade union or employee organization or group which may qualify as a trade union in respect of or affecting Employees or independent contractors nor is the Corporation subject to any union organization effort, nor is it engaged in any labour negotiation.
- (e) Schedule 2.1(20)(e) lists all material written employee benefit, fringe benefit, supplemental unemployment benefit, bonus, incentive, profit sharing, termination, change of control, pension, retirement, stock option, stock purchase, stock appreciation, health, welfare, medical, dental, disability, life insurance and similar plans, programs, arrangements or practices for the benefit of the current or former Employees, officers or directors of the Corporation that are currently maintained, sponsored or funded by the Corporation, whether funded or unfunded, insured or self-insured, registered or unregistered, other than plans established pursuant to statute (collectively, the "Employee Plans").
- (f) The Vendor has furnished to the Purchaser copies of all the Employee Plans, together with all related trust or other funding agreements, employee booklets and, where applicable, actuarial reports most recently filed with the applicable Governmental Authorities.
- (g) Except as would not reasonably be likely to result in a Material Adverse Change, each Employee Plan has been administered and funded in accordance with all Applicable Laws and the terms of the applicable Employee Plan.
- (h) No Employee Plan, nor any related trust or other funding medium thereunder, is subject to any pending or, to the knowledge of the Vendor, Threatened Legal Proceeding initiated by any Governmental Authority or by any other Person, other than routine claims for benefits.
- (21) Transactions with Affiliates et al. The Corporation is not liable in respect of advances, loans, guarantees, liabilities or other obligations to or on behalf of any shareholder, officer, director or Employee of the Corporation or any Affiliate of any of the foregoing. Except

as set forth on Schedule 2.1(21) and excluding access to the assets and the services provided to the Corporation by the Vendor or one of its Affiliates pursuant to the Transition Services Agreement, there are no intercompany services provided to the Corporation by the Vendor or by any Affiliate of the Vendor.

- (22) Tax Matters.
- (a) None of the Shares is "taxable Canadian property" within the meaning of the ITA. None of the Shares derives, and none of them has at any time within the past 60 months derived, more than 50% of its fair market value directly or indirectly from or from any combination of (i) real or immovable property situated in Canada, (ii) Canadian resource properties (as defined in the ITA), (iii) timber resource properties (as defined in the ITA), and (iv) options in respect of, or interests in, or for civil law, a right in, property described in any of (i) to (iii) above, whether or not the property exists.
- (b) The Corporation has prepared and filed when due with each relevant Governmental Authority all Tax Returns required to be filed by or on behalf of the Corporation in respect of any Taxes. All such Tax Returns are correct and complete in all material respects, and no material fact has been omitted therefrom. No extension of time in which to file any such Tax Returns is in effect. No Governmental Authority has asserted that the Corporation is required to file Tax Returns or pay any Taxes in any jurisdiction where it does not do so.
- (c) The Corporation has paid in full and when due all Taxes required to be paid by it, whether or not such Taxes are shown on a Tax Return or on any assessments or reassessments.
- (d) No assessments or reassessments of the Taxes of the Corporation are currently the subject of an objection or appeal, no audit by any Governmental Authority of the Corporation is currently ongoing and there are no outstanding issues which have been raised and communicated to the Corporation by any Governmental Authority. Neither the Vendor nor the Corporation has received any indication from any Governmental Authority that an audit, assessment or reassessment of the Corporation is proposed in respect of any Taxes, regardless of its merits. The Corporation has not executed or filed with any Governmental Authority any agreement or waiver extending the period for assessment, reassessment or collection of any Taxes.
- (e) Except as set forth on Schedule 2.1(22), the Corporation has withheld from each payment made to any Person, including any of its present or former Employees, officers and directors, and all Persons who are or are deemed to be non-residents of Canada for purposes of the ITA, all amounts required by Applicable Law to be withheld, and has remitted such withheld amounts within the prescribed periods to the appropriate Governmental Authority. The Corporation has remitted all Canada Pension Plan contributions, provincial pension plan contributions, employment insurance premiums, employer health taxes and other Taxes payable or required to be withheld and remitted by it in respect of the Employees to the appropriate Governmental Authority within the time required under Applicable Law. The Corporation has charged, collected and remitted on a

timely basis all Taxes as required under Applicable Law on any sale, supply or delivery whatsoever, made by the Corporation.

- (f) The terms and conditions made or imposed in respect of every transaction (or series of transactions) between the Corporation and any Person that is (i) a non-resident of Canada for purposes of the ITA, and (ii) not dealing at arm's length with the Corporation for purposes of the ITA, do not differ from those that would have been made between Persons dealing at arm's length for purposes of the ITA.
- (g) The Corporation has made or obtained records or documents that meet the requirements of paragraphs 247(4)(a) to (c) of the ITA with respect to all material transactions between it and any non-resident of Canada with whom it was not dealing at arm's length for purposes of the ITA.
- (h) There are no transactions or events that have resulted, and no circumstances existing which could result, in the application to the Corporation of sections 80, 80.01, 80.02, 80.03, 80.04 of the ITA or any analogous provision of any comparable law of any province or territory of Canada.
- (i) The Corporation has not incurred any deductible outlay or expense owing to a Person not dealing at arm's length (for purposes of the ITA) with the Corporation, the amount of which would, in the absence of an agreement filed under paragraph 78(1)(b) of the ITA, be included in the Corporation's income for Canadian income tax purposes for any taxation year or fiscal period beginning on or after the date hereof under paragraph 78(1)(a) of the ITA or any analogous provision of any comparable law of any province or territory of Canada.
- (23) No Conflicts. Except as set forth on Schedule 2.1(23), the execution and delivery and performance by the Vendor and the Corporation, as the case may be, of this Agreement (and each of the Closing Documents to which it is or is to become a party) do not and will not (with or without the giving of notice, the lapse of time or the happening of any other event or condition):
  - (a) result in the breach of, or conflict with, or allow any Person to exercise any rights under, or cause the Vendor or the Corporation to be bound by any additional or more onerous obligation under, any of the terms or provisions of:
    - (i) the articles, by-laws or any resolutions of the board of directors or shareholders of the Vendor or the Corporation; or
    - (ii) any material agreement, contract or commitment, to which the Vendor or the Corporation is a party, under which the Vendor or the Corporation has rights or obligations or by which any of the property or assets of the Vendor or Corporation may be affected, except, in the case of this clause (ii), for any such breach, conflict, or exercise of rights that would not reasonably be expected to be, individually or in the aggregate, material to the Corporation, taken as a whole or to prevent consummation of the Closing by the Vendor;

- (b) result in the breach of, or cause the termination, amendment or revocation of, any material consent or material License held by the Vendor or the Corporation or necessary to the ownership of the Shares or the operation of the Business;
- (c) result in the violation of any Applicable Law; or
- (d) result in the creation of any Lien (other than a Permitted Lien) on the Shares or any of the property or assets of the Corporation.
- (24) Required Regulatory Approvals. There is no requirement on the part of the Vendor or the Corporation to obtain any Regulatory Approval with any Governmental Authority in connection with the completion of the transactions contemplated by this Agreement other than those that may be required solely by reason of the Purchaser's (as opposed to any other third party's) participation in the transactions contemplated by this Agreement.
- (25) Required Consents. Except as set forth on Schedule 2.1(25), there is no requirement on the part of the Vendor or the Corporation to obtain any consent (other than a Regulatory Approval) in connection with the completion of the transactions contemplated by this Agreement to maintain all rights and benefits of the Corporation under any material Contract, Order or License after Closing except such consents which would not reasonably be expected to be, individually or in the aggregate, material to the Corporation, taken as a whole or prevent consummation of the Closing by the Vendor.
- (26) Dividends and Distributions. Except as set forth on Schedule 2.1(26), since the date of the Interim Financial Statements, the Corporation has not, directly or indirectly, declared or paid any dividends or declared or made any other distribution or return of capital in respect of any of its shares (or been deemed under the ITA to have done so) and has not, directly or indirectly, redeemed, purchased or otherwise acquired any of its outstanding shares or agreed to do so.
- (27) Business Carried on in Ordinary Course. Except as set forth on Schedule 2.1(27), since the date of the Interim Financial Statements, the Corporation has not:
  - (a) sold or otherwise disposed of any of its assets except for inventory sold in the ordinary course of business;
  - (b) incurred or assumed any obligation or liability of any nature whatsoever other than current liabilities incurred in the ordinary course of business;
  - (c) discharged any Lien or paid any secured or unsecured obligation or liability of any nature whatsoever, other than current liabilities incurred in the ordinary course of business, or scheduled payments under Contracts;
  - (d) created or permitted to exist any Lien on any of its assets other than a Permitted Lien;
  - (e) made any capital expenditure other than in the ordinary course of business:
  - (f) suffered any extraordinary loss, whether or not covered by insurance;

- (g) made any material change to the method of billing customers or the credit terms made available to customers;
- (h) made any material change to any method of management, operation or accounting in respect of the Business;
- (i) waived or cancelled any material rights or claims;
- (j) compromised or settled any Legal Proceeding pending against it, the Business or any of its assets;
- (k) written off any accounts receivable other than in the ordinary course of business;
- (I) hired or dismissed any Employee whose annual remuneration exceeds \$150,000;
- (m) increased the rate of wages, salaries or bonuses of any Employees except as required under the terms of any Contract;
- (n) increased the benefits to which Employees and former employees are entitled under any Employee Plan or created any new Employee Plan;
- (o) entered into, modified, amended or terminated any Contract or waived or released any rights under any Contract, other than in the ordinary course of business;
- (p) terminated, discontinued, closed or disposed of any plant, facility or business operation; or
- (q) authorized, agreed or become bound to do any of the foregoing.
- (28) Insurance. Schedule 2.1(28) contains a complete and accurate list and description of all insurance policies currently maintained by the Corporation (collectively, the "Insurance Policies"), including a brief description of the type of insurance, the name of the insurer, policy number, coverage limits and term. True and complete copies of the Insurance Policies have been made available to the Purchaser for inspection. Each of the Insurance Policies is with a reputable and sound insurer and covers all of the Corporation's property and assets and protects the Business in such amounts and against such losses and claims as is adequate for the conduct of its businesses and the value of its properties and as is customary for companies engaged in comparable businesses with similar properties and in similar industries. Each of the Insurance Policies is valid and subsisting and in good standing and there is no default under any of them.
- **2.2** Representations and Warranties of the Purchaser. The Purchaser represents and warrants to the Vendor, except as set forth in the Schedules, as of the date hereof, as follows.
- (1) Incorporation and Corporate Power. The Purchaser is a corporation incorporated, organized and subsisting under the laws of the jurisdiction of its incorporation. The Purchaser has the corporate power, authority and capacity to execute and deliver this Agreement and all other agreements and instruments to be executed by it as contemplated herein and to perform its obligations under this Agreement and under all such other agreements

and instruments to be executed by it as contemplated herein and to perform its obligations hereunder and under all such other agreements and instruments.

- (2) Authorization and Enforceability. The execution, delivery and performance by the Purchaser of this Agreement and all agreements and instruments to be executed and delivered by the Purchaser hereunder have been duly authorized by all necessary corporate action on the part of the Purchaser and no other corporate actions on the part of the Purchaser are necessary to authorize the execution, delivery or performance by the Purchaser of this Agreement or such other agreements and instruments, and this Agreement, assuming due and valid authorization, execution and delivery hereof by the Vendor, constitutes the valid and binding obligation of the Purchaser enforceable against the Purchaser in accordance with its terms, except as limited by the application of bankruptcy, insolvency, reorganization, fraudulent conveyance, moratorium or other Applicable Laws relating to or affecting creditors' rights or general principles of equity.
- (3) Legal Proceedings. As of the date hereof, there are no Legal Proceedings pending or, to the knowledge of Purchaser, Threatened against Purchaser at law or in equity, or before or by any Governmental Authority, except for any such Legal Proceedings that have not had and would not reasonably be expected to have, individually or in the aggregate, a material and adverse effect on the ability of the Purchaser to consummate the transactions contemplated by this Agreement.
- (4) No Conflict. The execution and delivery and performance by the Purchaser of this Agreement (and each of the Closing Documents to which it is or is to become a party) do not and will not (with or without the giving of notice, the lapse of time or the happening of any other event or condition):
  - (a) result in the breach of, or conflict with, or allow any Person to exercise any rights under, or cause the Purchaser to be bound by any additional or more onerous obligation under, any of the terms or provisions of:
    - (i) the articles, by-laws or any resolutions of the board of directors or shareholders of the Purchaser; or
    - (ii) any material agreement, contract or commitment, to which the Purchaser is a party, under which the Purchaser has rights or obligations or by which any of the property or assets of the Purchaser may be affected, except, in the case of this clause (ii), for any such breach, conflict, or exercise of rights that would not reasonably be expected to be, individually or in the aggregate, material to the Purchaser, taken as a whole or to prevent consummation of the Closing by the Purchaser;
  - (b) result in the breach of, or cause the termination, amendment or revocation of, any material consent or material License held by the Purchaser or necessary to the ownership of the Shares or the operation of the Business;
  - (c) result in the violation of any Applicable Law; or
  - (d) result in the creation of any Lien on any of the property or assets of the Corporation.

- (5) Investigation.
- (a) To the knowledge of Purchaser, there is no fact relating to Purchaser's or any of its Affiliates' businesses, operations, financial condition or legal status that would or would reasonably be expected to: (i) prevent or prohibit the obtaining of, impose any material delay in the obtaining of or increase the risk of not obtaining any consent or Order of, or filing with, any Governmental Authority necessary to consummate the transactions contemplated by this Agreement or the expiration or termination of any applicable waiting period (and any extension thereof) or (ii) prohibit, prevent or materially delay the consummation of the transactions contemplated by this Agreement.
- (b) Purchaser acknowledges that Purchaser is relying on its own investigation and analysis in entering into the transactions contemplated by this Agreement. Purchaser is knowledgeable about the industries in which the Corporation operates and is capable of evaluating the merits and risks of the transactions contemplated by this Agreement and is able to bear the substantial economic risk of such investment for an indefinite period of time. Purchaser has been afforded full access to the Books and Records, facilities and personnel of the Corporation for purposes of conducting a due diligence investigation, has received all information requested by it and its Representatives from the Corporation and has conducted a full due diligence investigation of the Corporation.
- (6)Investment Intent; Restricted Securities. Purchaser is acquiring the Shares solely for Purchaser's own account, for investment purposes only and not with a view to, or with any present intention of, reselling or otherwise distributing the Shares or dividing its participation therein with others. Purchaser understands and acknowledges that (i) none of the Shares have been registered or qualified under any securities laws of any state of the United States, province or territory of Canada or any other jurisdiction and were issued in reliance on specific exemptions thereunder, (ii) the Shares constitute "restricted securities" as defined in Rule 144 under the Securities Act of 1933, as amended (the "Securities Act"), (iii) none of the Shares are traded or tradable on any securities exchange or over the counter and (iv) none of the Shares may be sold, transferred, offered for sale, pledged, hypothecated or otherwise disposed of unless a registration statement under the Securities Act with respect to such Shares and qualification in accordance with any applicable state securities laws becomes effective or a prospectus exemption under Canadian securities laws or unless such registration and qualification is inapplicable, or an exemption therefrom is available. Purchaser will not sell, transfer, offer for sale, pledge, hypothecate or otherwise dispose of any of the Shares acquired hereunder or any interest therein in any manner that may cause Vendor to be in violation of the Securities Act or any applicable securities laws. Purchaser is an "accredited investor" as defined in Rule 501(a) of the Securities Act and in National Instrument 46-106 - Prospectus Exemptions. Purchaser acknowledges that it is informed as to the risks of the transactions contemplated by this Agreement and of ownership of the Shares.
  - (7) Solvency; Fraudulent Conveyance.
  - (a) Assuming that (i) the representations and warranties concerning the Corporation contained in this Agreement are true and correct in all material respects, and (ii) the Corporation is solvent immediately prior to the Closing, as of the Closing, immediately after giving effect to all of the transactions contemplated by this Agreement the Corporation will be solvent.

- (b) No transfer of property is being made and no obligation is being incurred in connection with the transactions contemplated by this Agreement with the intent to hinder, delay or defraud either present or future creditors of the Vendor or the Corporation.
- (8) Required Regulatory Approvals. There is no requirement on the part of the Purchaser or the Corporation to obtain any Regulatory Approval with any Governmental Authority in connection with the completion of the transactions contemplated by this Agreement other than those that may be required solely by reason of the Vendor's (as opposed to any other third party's) participation in the transactions contemplated by this Agreement.
- **2.3 Commissions**. Each Party represents and warrants to the other Party that such other Party will not be liable for any brokerage commission, finder's fee or other similar payment in connection with the transactions contemplated by this Agreement because of any action taken by, or agreement or understanding reached by, that Party.
- 2.4 No Additional Representations or Warranties. Except for the representations and warranties of the Vendor and the Corporation expressly set forth in this Article 2 regarding the Vendor and the Corporation, as applicable, in each case, as qualified by the Schedules and in accordance with the express terms and conditions (including limitations and exclusions) of this Agreement (the "Express Representations") (it being understood that the Purchaser has relied only on such Express Representations), the Purchaser acknowledges and agrees, on its own behalf and on behalf of any Purchaser's Indemnified Party, that neither the Corporation nor any other Person on behalf of the Corporation makes, and the Purchaser has not relied on, and is not relying on, the accuracy or completeness of any express or implied representation or warranty with respect to the Corporation or with respect to any statement or information of any nature made or provided by any Person, any information, statements, disclosures, documents, projections, forecasts or other material made available to the Purchaser or any of its Affiliates or Representatives in that certain datasite "Project Seitel" administered by Intralinks (the "Dataroom"), or the projections on behalf of the Corporation or any of its Affiliates or Representatives to the Purchaser or any of its Affiliates or Representatives. Without limiting the foregoing, neither the Corporation nor any other Person will have or be subject to any liability whatsoever to the Purchaser, or any other Person, resulting from the distribution to the Purchaser or any of its Affiliates or Representatives, or the Purchaser or any of its Affiliates' or Representatives' use of or reliance on, any such information, any information, statements, disclosures, documents, projections, forecasts or other material made available to the Purchaser or any of its Affiliates or Representatives in the Dataroom or otherwise provided to the Purchaser in expectation of the transactions or any discussions with respect to any of the foregoing information.

### Article 3 SURVIVAL AND INDEMNIFICATION

**3.1 Survival**. The representations and warranties of the Vendor and the Purchaser contained in or made pursuant to this Agreement or in any certificate furnished pursuant to this Agreement shall survive the Closing and terminate on the date that is twelve (12) months after the date hereof (and no claims shall be made for indemnification with respect thereto under Sections 3.2 or 3.3 thereafter). The covenants and agreements that by their terms apply or are to be performed in whole or in part after the date hereof shall survive the Closing and remain in effect for the period provided in such covenants and agreements, if any, or if no such period is provided, until the first anniversary of the date hereof. The indemnification set forth in

Sections 3.2(c) and 3.2(d) shall terminate on the date that is twelve (12) months after the date hereof (and no claims shall be made for indemnification pursuant to (i) Section 3.2(d) thereafter and (ii) Sections 3.2(c) five (5) days thereafter).

- **3.2** Indemnity by the Vendor. Subject to this Article 3, the Vendor shall indemnify the Purchaser's Indemnified Parties and will reimburse them for, any actual out-of-pocket losses, liabilities, damages or expenses (including reasonable attorneys' fees and disbursements) ("Damages") arising from or related to:
  - (a) any breach of any representation or warranty of the Vendor contained in this Agreement or in any other agreement, certificate or instrument executed and delivered to the Purchaser pursuant to this Agreement:
  - (b) any breach of any covenant or agreement on the part of the Vendor contained in this Agreement or in any other agreement, certificate or instrument executed and delivered to the Purchaser pursuant to this Agreement;
  - (c) any Corporation Liabilities that are Finally Determined to be in excess of \$4,114,803; and
  - (d) any Taxes required to be paid by the Corporation for (i) any Pre-Closing Tax Period, or (ii) the portion of a Straddle Period ending on the date hereof.
- **3.3** Indemnity by the Purchaser. Subject to this Article 3, the Purchaser shall indemnify the Vendor's Indemnified Parties and will reimburse them for, any Damages arising from or related to:
  - (a) any breach of any representation or warranty of the Purchaser contained in this Agreement or in any other agreement, certificate or instrument executed and delivered pursuant to this Agreement; and
  - (b) any breach or non-fulfilment of any covenant or agreement on the part of the Purchaser contained in this Agreement or in any other agreement, certificate or instrument executed and delivered pursuant to this Agreement.
- **3.4 Claim Notice**. If an Indemnified Party becomes aware of any act, omission or state of facts that may give rise to Damages in respect of which a right of indemnification is provided for under this Article 3, the Indemnified Party shall promptly give written notice thereof (a "**Claim Notice**") to the Indemnifying Party. The Claim Notice shall specify whether the potential Damages arise as a result of a claim by a Person against the Indemnified Party (a "**Third Party Claim**") or whether the potential Damages arise as a result of a claim directly by the Indemnified Party against the Indemnifying Party (a "**Direct Claim**"), and shall also specify with reasonable particularity (to the extent that the information is available):
  - (a) the factual and contractual basis for the Direct Claim or Third Party Claim, as the case may be; and
  - (b) the amount of the potential Damages arising therefrom, if known.

If, through the fault of the Indemnified Party, the Indemnifying Party does not receive a Claim Notice in time effectively to contest the determination of any liability susceptible of being

contested or to assert a right to recover an amount under applicable insurance coverage, then the liability of the Indemnifying Party to the Indemnified Party under this Article 3 shall be reduced only to the extent that the Indemnifying Party is actually prejudiced by such delay in time.

- **3.5 Monetary Limitations**. No Damages may be recovered from the Vendor pursuant to Section 3.2(a):
  - (a) in respect to any individual item (or group of related items) unless the accumulated amount of Damages of the Purchaser's Indemnified Parties related thereto exceeds \$50,000 the ("Per Claim Threshold"); and
  - (b) unless and until the aggregate amount of Damages of the Purchaser's Indemnified Parties arising pursuant to Section 3.2(a) exceeds \$630,000 (the "Threshold"),

in which event the aggregate amount of only those Damages in excess of the Threshold may be recovered. In no event shall Damages be recovered by the Purchaser:

- (c) pursuant to Section 3.2(a) (other than in respect of breaches of Section 2.1(1), Section 2.1(2), Section 2.1(3), Section 2.1(4), Section 2.1(5) and Section 2.1(6)) and Section 3.2(d) in excess of \$7,950,000,
- (d) pursuant to Section 3.2(c) in excess of the then remaining Escrow Amount, and
- (e) pursuant to Section 3.2 in excess of the Purchase Price.

The Escrow Amount shall be used for the sole purpose of, and shall be the sole source for, satisfying any Damages for which the Purchaser's Indemnified Parties are entitled to indemnification pursuant to Section 3.2(c).

#### 3.6 Certain Additional Matters.

- (1) The amount of any and all Damages under this Article 3 will be determined net of (i) any amounts actually recovered by an Indemnified Party or any of such Indemnified Party's Affiliates under or pursuant to any insurance policy, title insurance policy, indemnity, reimbursement arrangement or contract pursuant to which or under which such Indemnified Party or such Indemnified Party's Affiliates is a party or has rights (collectively, "Alternative Arrangements") and (ii) the amount of any Tax benefit available to an Indemnified Party or any of such Indemnified Party's Affiliates (or their direct or indirect beneficial owners) with respect to such Damages.
- (2) No Indemnified Party shall have the right to recover under Section 3.2 with respect to any Damages or alleged Damages to the extent the matter forming the basis for such Damage or alleged Damage shall have been taken into account in the calculation of the \$4,114,803 of Corporation Liabilities set forth in Section 3.2(c) or the Closing Working Capital.
- (3) No claim shall be made with respect to Damages arising out of any breach, or facts constituting a breach, of this Agreement actually known by any Indemnified Party to exist at or prior to the date hereof. Any Damage for which any Indemnified Party is entitled to indemnification under Section 3.2 shall be determined without duplication of recovery by reason

of the state of facts giving rise to such Damage constituting a breach of more than one representation, warranty, covenant or agreement or constituting a Corporation Liability or Tax. No Indemnified Party shall have the right to assert any claim pursuant to this Article 3 with respect to any Damage, cause of action or other claim to the extent it is primarily a possible or potential Damage, cause of action or claim that the Indemnified Party believes may be asserted rather than an actual Damage, cause of action or claim that has, in fact, been filed of record or asserted against any Indemnified Party, paid or incurred by any Indemnified Party.

- In no event will any Indemnified Party be entitled to recover or make a claim for (4) any amounts in respect of, and in no event will "Damages" be deemed to include, (i) punitive, special, consequential or exemplary damages (except to the extent actually paid to a third party) and, in particular, no "diminution of value," "multiple of profits" or "multiple of cash flow" or similar valuation methodology shall be used in calculating the amount of any Damages, (ii) any liability arising from any change after the date hereof in any Applicable Law or any interpretation thereof, (iii) any Damage arising from reliance after the Closing on any facts represented relating to Taxes, Tax Returns or related matters prior to the Closing continuing to be true after the Closing or (iv) any Taxes (I) of Purchaser or any of its subsidiaries (other than the Corporation), (II) with respect to any Taxable period (or portion thereof) beginning on or after the date hereof, (III) that are due to the unavailability in any taxable period (or portion hereof) beginning after the date hereof of any net operating loss, credit or other Tax attribute from a Pre-Closing Tax Period (or portion thereof), (IV) that are due to transactions occurring on the date hereof but after the Closing, (V) that are due to the manner in which Purchaser finances the transactions contemplated by this Agreement or (VI) that are due to Purchaser's breach of Section 4.1.
- (5) To the extent permitted by Applicable Law, any indemnity payment under this Agreement will be treated as an adjustment to the Purchase Price for Tax purposes.
- (6) Each Indemnified Party agrees that in the event of any breach giving rise to an indemnification obligation under this Article 3, such Indemnified Party will take and will cause its Affiliates to take, or cooperate with the Indemnifying Party, if so requested by the Indemnifying Party, in order to take, all reasonable measures to mitigate the consequences of the related breach (including taking steps to prevent any contingent liability from becoming an actual liability).
- (7) No Purchaser Indemnified Party shall be entitled to recover or make a claim for any amounts in respect of the Corporation Liabilities pursuant to Section 3.2(c) unless and until (i) with respect to the Office Lease and Warehouse Lease (each as defined on Schedule 4.10), the Purchaser has provided Vendor evidence of satisfaction of the Purchaser's obligations in Section 4.10 in a form reasonably satisfactory to Vendor, and (ii) with respect to Employee Severance (as defined on Schedule 4.10), the Purchaser has provided Vendor evidence of such payments in a form reasonably satisfactory to Vendor.
- **3.7 Direct Claims.** In the case of a Direct Claim, the Indemnified Party shall make available to the Indemnifying Party the information relied upon by the Indemnified Party to substantiate its right to be indemnified under this Article 3, together with all such other information as the Indemnifying Party may reasonably request.
- **3.8 Third Party Claims**. In the event a Claim Notice is delivered timely with respect to a Third Party Claim, the Indemnifying Party may assume the defense and control of such Third Party Claim by delivery of written notice to the Indemnified Party; provided, that the

Indemnifying Party shall not be entitled to assume or maintain control of the defense of any Third Party Claim to the extent the Third Party Claim primarily seeks non-monetary relief or involves criminal allegations. The assumption of the defense by the Indemnifying Party of any Third Party Claim shall not require the Indemnifying Party to agree to be liable for any Damages in respect of such Third Party Claim and shall be without prejudice to any rights or defenses of the Indemnifying Party in respect of whether the Indemnified Party is entitled to indemnification under this Article 3 for any particular Damages. Should an Indemnifying Party elect to assume the defense of a Third Party Claim, the Indemnifying Party will not be liable to the Indemnified Party for legal expenses subsequently incurred by the Indemnified Party in connection with the defense thereof. If the Indemnifying Party assumes such defense, the Indemnified Party will have the right to participate in the defense thereof and to employ counsel, at its own expense, separate from the counsel employed by the Indemnifying Party, it being understood, however, that the Indemnifying Party will control such defense. If the Indemnifying Party chooses to defend any Third Party Claim, all the Parties will cooperate in the defense or prosecution of such Third Party Claim. Such cooperation will include the retention and (upon the Indemnifying Party's request) the provision to the Indemnifying Party of records and information reasonably relevant to such Third Party Claim and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. Whether or not the Vendor has assumed the defense of a Third Party Claim, neither Purchaser nor any of its Affiliates (including the Corporation) will admit any liability with respect to, consent to the entry of any judgment, or settle, compromise or discharge, any Third Party Claim without the prior written consent of the Vendor.

- **3.9 Manner of Payment**. Subject to, and in accordance with, the provisions of this Article 3 and the Escrow Agreement, following a Final Determination:
- (1) Any indemnification of the Vendor's Indemnified Parties pursuant to this Article 3 shall be delivered by the Purchaser to the applicable Vendor's Indemnified Parties by wire transfer of immediately available funds to an account designated by the Vendor within five (5) Business Days after the determination thereof.
- (2) Any indemnification of the Purchaser's Indemnified Parties pursuant to Section 3.2(c) shall be delivered to the Purchaser from the Escrow Account in accordance with the terms of the Escrow Agreement by wire transfer of immediately available funds to an account designated by the Purchaser within five (5) Business Days after the determination thereof.
- (3) Subject to the terms of the Escrow Agreement, on the date that is five days after the first anniversary of the date hereof, the Vendor and the Purchaser shall deliver to the Escrow Agent a joint written instruction directing the Escrow Agent to release to the Vendor the remaining portion of the Escrow Amount (including any accrued interest), less an aggregate amount equal to the total of all claims for indemnification pursuant to Section 3.2(c) that are properly asserted and pending pursuant to this Article 3 prior to such date (which aggregate amount shall be released in accordance with this Article 3).
- **3.10 Exclusive Remedy**. From and after the Closing, the Purchaser's Indemnified Parties' sole and exclusive remedy against the Vendor and its Affiliates, whether in any individual, corporate or any other capacity, with respect to any and all claims relating (directly or indirectly) to the subject matter of this Agreement or the transactions contemplated hereby, regardless of the legal theory under which such liability or obligation may be sought to be imposed, whether sounding in contract or tort, or whether at law or in equity, or otherwise, shall be pursuant to the provisions of this Article 3 and the provisions of the Escrow Agreement. In furtherance of the

foregoing, the Purchaser hereby waives and releases to the fullest extent permitted under Applicable Law, the Corporation, the Vendors and its Affiliates, whether in any individual, corporate or any other capacity, from and against any and all other rights, claims and causes of action it may have against the Corporation, the Vendor or its Affiliates relating (directly or indirectly) to the subject matter of this Agreement or the transactions contemplated hereby (including relating to any Exhibit, Schedule or document delivered hereunder), including whether arising under or based upon any Applicable Law or otherwise, other than in the case of Fraud. Except for the Purchaser pursuant to Section 1.10(2), no Person (including the Vendor and its Affiliates) shall have any obligation to fund the Escrow Account. The provisions in this Agreement relating to indemnification, and the limits imposed on the Purchaser's Indemnified Parties' remedies with respect to this Agreement and the transactions contemplated hereby (including Section 3.2) were specifically bargained for between sophisticated parties and were specifically taken into account in the determination of the amounts to be paid to the Vendor hereunder. No Purchaser's Indemnified Party may avoid the limitations on liability set forth in this Article 3 by seeking Damages for breach of contract, tort or pursuant to any other theory of liability. Nothing in this Section 3.10 shall limit the rights of the Corporation and the Vendor to seek specific performance of the other Parties' obligations hereunder in accordance with Section 5.18.

### Article 4 COVENANTS

#### 4.1 Preparation of Tax Returns.

- Cooperation. The Vendor shall cause to be prepared and filed on a timely basis all Tax Returns for the Corporation for (i) any Pre-Closing Tax Period for which Tax Returns have not been filed as of the date hereof and (ii) for any Straddle Period for which Tax Returns are required to be prepared and filed (all Tax Returns referred to in clause (i) and (ii) above collectively being referred to herein as the "Stub Period Returns"). The Vendor and the Purchaser shall co-operate fully with each other and make available to each other in a timely fashion such data and other information as may reasonably be required for the preparation of all Stub Period Returns and shall preserve such data and other information until the expiration of any applicable limitation period under any Applicable Law with respect to such Stub Period Returns. The Vendor shall provide to the Purchaser for its review a copy of the Stub Period Returns no later than 45 days in the case of an income Tax Return, and 15 days in the case of any other Tax Return, prior to the due date for filing such Tax Return with the appropriate Governmental Authorities. The Purchaser shall notify the Vendor in writing within 30 days in the case of an income Tax Return, and 10 days in the case of any other Tax Return, after delivery to it of a Stub Period Return if it has any comments with respect to items set forth in such Stub Period Return, including the substance of its comments. Notwithstanding the foregoing, each Stub Period Return shall, at the request of the Vendor, contain (a) a request to carry back losses, credits or other available tax attributes to such earlier taxation years or periods as are specified by the Vendor in its sole discretion, and/or (b) an election by the Corporation pursuant to subsection 256(9) of the ITA in respect of its taxation year ending immediately before the acquisition of control of it by the Purchaser.
- (2) Acknowledgement. Notwithstanding anything herein to the contrary, the Parties acknowledge and agree that for purposes of determining the items of income, gain, loss or deduction with respect to the transfer or distribution by the Corporation of the assets set forth on Schedule 4.1 (the "**Transferred Assets**") prior to the Closing, the value of such Transferred

Assets shall be \$10,837,575.69 The Purchaser will cause the Corporation to timely remit any Taxes shown as owing on such Stub Period Returns.

- (3) Post-Closing Actions. Without the prior written consent of the Vendor, neither the Purchaser nor any of the Purchaser's Affiliates (including the Corporation) shall (or shall cause or permit any other Person to) (i) except as provided in Section 4.1 amend, file, re-file or otherwise modify any Tax Return relating in whole or in part to the Corporation with respect to any Pre-Closing Tax Period or the pre-Closing portion of any Straddle Period, (ii) make any Tax election that has retroactive effect to any Pre-Closing Tax Period or the pre-Closing portion of any Straddle Period, (iii) file a ruling request with any Governmental Authority that could affect the Taxes or Tax Returns of the Corporation or the Vendor for a Pre-Closing Tax Period or the pre-Closing portion of any Straddle Period, (iv) agree to waive or extend the applicable statute of limitations with respect to any Tax matter involving the Corporation that relates to any Pre-Closing Tax Period or the pre-Closing portion of any Straddle Period or (v) pursue or enter into any voluntary disclosure with any Governmental Authority regarding any Tax or Tax Returns of the Corporation for a Pre-Closing Tax Period or the pre-Closing portion of any Straddle Period.
- Audit of the Corporation. The Purchaser acknowledges that the Vendor has, prior to 4.2 the date hereof, engaged BKD LLP (the "Auditor") to perform an audit of the financial statements of the Corporation of the fiscal years ending December 31, 2017 (the "2017 Audit") and December 31, 2018 (the "2018 Audit") prepared in accordance with U.S. generally accepted accounting principles. From and after the date hereof, the Vendor shall (i) continue to provide information with respect to the assets and business of the Corporation, and shall facilitate the preparation and audit of the financial statements of the Corporation for the fiscal years ending December 31, 2017 and December 31, 2018, including meeting with internal accounting and auditing personnel and external accountants and auditors and (ii) execute such customary certifications and other instruments as reasonably requested by external accountants and auditors in connection with the audit of such financial statements to the extent such certifications and other instruments relate to periods prior to the date hereof and are consistent with past practice. In furtherance of the foregoing, Purchaser shall cooperate with the Vendor, including by providing the Vendor and its Representatives full access to the books, records, supporting data, facilities and personnel of the Corporation (including the Vendor personnel responsible for accounting and finance, senior management and the Corporation's accountants and other Representatives) for purposes of completing the 2017 Audit and 2018 Audit. The Vendor shall use its commercially reasonable best efforts to cause the Auditor to deliver the 2017 Audit and the 2018 Audit prior to the date on which Purchaser is required to file a business acquisition report in respect of the transactions contemplated hereby; provided that any delay caused by the Purchaser's failure to provide timely information requested by the Vendor pursuant to, or other failure to comply with, this paragraph shall not be deemed a breach of the Vendor's obligations hereunder. All fees and expenses of the Auditor with respect to the 2017 Audit shall be borne by the Vendor and with respect to the 2018 Audit shall be borne by Purchaser.

#### 4.3 Director and Officer Liability.

(1) For at least six (6) years following the date hereof, the Purchaser will not, and will not permit the Corporation to, amend, repeal or modify in any manner adverse to any present (as of immediately prior to the Closing) and former officer, director, supervisory director, employee, manager, managing member, member, partner (general or limited), fiduciary or agent of the Corporation (each, an "Indemnified Person") any provision in such Person's certificates or articles of incorporation, articles of organization, certificate of limited partnership, operating

agreement, bylaws, limited partnership agreement or similar governing documents, as applicable, or in any agreement, relating to the exculpation or indemnification of, or advancement of expenses to, any Indemnified Person as in effect immediately prior to the Closing, and the Purchaser will cause all such provisions to be observed by the Corporation, it being the intent of the Parties that any Indemnified Person will continue to be entitled to such exculpation, indemnification and advancement of expenses to the fullest extent permitted under Applicable Law.

- (2) In the event that the Purchaser or the Corporation or any of the respective successors or assigns of the foregoing (i) consolidates with, merges into or amalgamates with any other Person or (ii) transfers all or substantially all of its properties or assets to any Person, then, in each case, the successors and assigns of such Persons or properties or assets, as the case may be, will, and the Purchaser and the Corporation will cause such successors to, expressly assume in writing and be bound by the obligations set forth in this Section 4.3 as a condition of succession of assignment.
- (3) This Section 4.3 is intended to be for the benefit of each of the Indemnified Persons and may be enforced by any such Indemnified Person as if such Indemnified Person were a party to this Agreement. The obligations of the Purchaser under this Section 4.3 will not be terminated or modified in such a manner as to adversely affect any Person to whom this Section 4.3 applies without the consent of such affected Person.
- 4.4 Access to Books and Records. From and after the Closing for a period of seven (7) years following the date hereof, (i) the Purchaser will, and will cause the Corporation to, provide the Vendor and its Representatives with reasonable access, during normal business hours, and upon reasonable advance notice, to the Books and Records, including work papers, schedules, memoranda, Tax Returns, Tax schedules, Tax rulings, and other documents (for the purpose of examining and copying) of the Corporation with respect to periods or occurrences prior to the date hereof and reasonable access, during normal business hours, and upon reasonable advance notice, to employees, officers, advisors, accountants, offices and properties (including for the purpose of better understanding the Books and Records) of the Purchaser and the Corporation and (ii) the Vendor will provide the Purchaser and its Representatives with reasonable access, during normal business hours, and upon reasonable advance notice, to the Books and Records, including work papers, schedules, memoranda, Tax Returns, Tax schedules, Tax rulings, and other documents (for the purpose of examining and copying) of the Corporation with respect to periods or occurrences prior to the date hereof.
- **4.5 Conflicts.** Each of the Parties hereby agrees, on its own behalf and on behalf of its directors, members, partners, managers, members, officers, employees and Affiliates (including, in the case of the Purchaser after the Closing, the Corporation), that Kirkland & Ellis LLP may serve as counsel to each and any of the Vendor and its controlling Affiliates (individually and collectively, the "**Vendor Group**"), on the one hand, and the Corporation, on the other hand, in connection with the negotiation, preparation, execution, delivery and performance of this Agreement, and the consummation of the transactions contemplated hereby, Kirkland & Ellis LLP (or any successor) may serve as counsel to the Vendor Group or any director, member, partner, manager, officer, employee or Affiliate of any member of the Vendor Group, in connection with any litigation, claim or obligation arising out of or relating to this Agreement or the transactions contemplated by this Agreement, notwithstanding such representation or any continued representation of the Corporation, and each of the Parties (on their own behalf and on behalf of their Affiliates (including, in the case of the Purchaser after the Closing, the Corporation)) hereby consents

thereto and irrevocably waives any conflict of interest arising therefrom, and each of such Parties shall cause any Affiliate thereof to consent to irrevocably waive any conflict of interest arising from such representation. In addition, (i) all communications between any member of the Vendor Group or any of their respective Affiliates, directors, officers, employees or Representatives, on the one hand, and Kirkland & Ellis LLP, on the other hand, made in connection with the negotiation, preparation, execution, delivery and Closing under, or any dispute arising in connection with, this Agreement, or otherwise relating to any potential sale of the Corporation (the "**Protected Vendor Communications**"), shall be deemed to be privileged and confidential communications, and (ii) all rights to such Protected Vendor Communications, and the control of the confidentiality and privilege applicable thereto, shall be retained by the Vendor Group.

- **4.6 Use of Name**. Within fifteen (15) days following Closing, the Purchaser shall cause the Corporation to change its names to a name that does not constitute, contain, or bear any resemblance to "Seitel" or any other business name used by or trademark belonging to the Vendor or any combination, derivation, adaptation, or translation of any of the foregoing (collectively, the "**Vendor Trademarks**"). Following such date, the Purchaser shall not (and shall cause each of its Affiliates, including the Corporation, not to) use any business name or trademark that constitutes, contains, or bears any resemblance to any of the Vendor Trademarks in the operation of their businesses.
- **4.7 Tax Refunds**. Any refunds or credits of Taxes (including any interest paid or credited with respect thereto, net of any Taxes paid by the Corporation on such interest) of, or with respect to, the Corporation for any Pre-Closing Tax Period or the pre-Closing portion of any Straddle Period will be for the account of the Vendor (except to the extent such refunds or credits were reflected in A/R Schedule). The Purchaser shall promptly inform the Vendor of any such refunds or credits to which the Vendor may be entitled hereunder and shall pay to the Vendor an amount equal to the amount of any such refunds or credits within 10 days following the date such refunds or credits were paid or credited by the relevant Governmental Authority to the Corporation. Any payment made pursuant to this Section 4.7 shall constitute an increase to the Purchase Price.

#### 4.8 Releases.

Effective upon the Closing, the Purchaser on behalf of itself, its Affiliates and the Corporation (collectively, the "Purchaser Releasers"), hereby knowingly, willingly, irrevocably and expressly waives, acquits, remises, discharges and forever releases the Vendor and each equityholders, officers, directors, members, managers, employees, Representatives and Affiliates (collectively, the "Vendor Released Parties"), from any and all liabilities and obligations to such Purchaser Releasers of any kind or nature whatsoever, whether in the capacity as an equityholder of the Corporation or otherwise, in each case whether absolute or contingent, liquidated or unliquidated, known or unknown, matured or unmatured or determined or determinable, and whether arising under any Applicable Law, contract, agreement, arrangement, commitment, undertaking or understanding, whether written or oral (other than this Agreement and any of the other agreements executed and delivered in connection herewith, but, in each case, only to the extent set forth herein or therein) or otherwise at law or in equity, and each of the Purchaser Releasers hereby agrees that it will not seek to recover any amounts in connection therewith or thereunder from any of the Vendor Released Parties (except as provided for in this Agreement or any of the other agreements executed and delivered in connection herewith, but, in each case, only to the extent set forth herein or therein).

Effective upon the Closing, the Vendor on behalf of itself, its Affiliates and each equityholders, officers. directors, members, managers, employees, Representatives and Affiliates (collectively, the "Vendor Releasers"), hereby knowingly, willingly, irrevocably and expressly waives, acquits, remises, discharges and forever releases the Purchaser, from any and all liabilities and obligations to such Vendor Releasers of any kind or nature whatsoever, in each case whether absolute or contingent, liquidated or unliquidated, known or unknown, matured or unmatured or determined or determinable, and whether arising under any Applicable Law, contract, agreement, arrangement, commitment, undertaking or understanding, whether written or oral (other than this Agreement and any of the other agreements executed and delivered in connection herewith, but, in each case, only to the extent set forth herein or therein) or otherwise at law or in equity, and each of the Vendor Releasers hereby agrees that it will not seek to recover any amounts in connection therewith or thereunder from the Corporation (except as provided for in this Agreement or any of the other agreements executed and delivered in connection herewith, but, in each case, only to the extent set forth herein or therein).

#### 4.9 Bonuses; Severance.

- (1) The Purchaser shall cause the Corporation to (i) honor and maintain the bonus programs for the Employees as of the date hereof and set forth under the heading "Bonus Programs" on Schedule 2.1(20)(e) until such time as the bonuses referred to in clause (ii) are paid and (ii) pay to the Employees the bonuses such Employees have accrued under such programs (with bonuses accrued as of the Closing Date to be taken into account in determining Current Liabilities) at the end of the bonus determination period that includes the Closing Date.
- (2) The Purchaser shall cause the Corporation to (i) honor and maintain the severance policy for the Employees as of the date hereof as set forth under the heading "Employee Severance" on Schedule 4.10 until such time as the Severance Pay (as defined in Schedule 4.10) referred to in clause (ii) is paid and (ii) pay to each Employee the Severance Pay payable to such Employee upon the termination of such Employee's employment with the Corporation in accordance with the severance policy.
- **4.10 Corporation Liabilities**. From and after the Closing, the Purchaser shall, and shall cause its Affiliates (including the Corporation) to, use reasonable best efforts to:
- (1) sublease the property (the "Office Space") associated with that certain Lease Agreement, made as of May 11, 2015, between Manufacturers Life Insurance Company, a body corporate, having its head office in Toronto, Canada, and having a local office at Suite 400, 550 6th Avenue S.W. in the City of Calgary, Province of Alberta and the Corporation for the period March 30, 2019 until November 30, 2022 in an amount equal to or greater than, on a monthly basis, [redacted commercially sensitive information]% of the monthly lease payments and operating expenses of the Office Space;
- (2) continue to collect all fees and expenses due and payable to the Corporation pursuant to section 5.2 of that certain Seismic Data Purchase Agreement, effective February, 1999, by and between [redacted confidential information], an Alberta Corporation and [redacted confidential information], an Alberta Corporation, and [redacted confidential information], a general partnership carrying on business in the Province of Alberta, as in effect on the date hereof, through January 30, 2020 and shall not take any action that would be reasonably expected to result in a decrease of such fees and expenses due and payable to the Corporation; and

(3) mitigate all costs associated with the Corporation Liabilities.

## Article 5 GENERAL

- **5.1** Actions on Non-Business Days. If any payment is required to be made or other action (including the giving of notice) is required to be taken pursuant to this Agreement on a day which is not a Business Day, then such payment or action shall be considered to have been made or taken in compliance with this Agreement if made or taken on the next succeeding Business Day.
- **5.2 Currency and Payment Obligations**. Except as otherwise expressly provided in this Agreement:
  - (a) all dollar amounts referred to in this Agreement are stated in Canadian Dollars; and
  - (b) any payment contemplated by this Agreement shall be made in cash, by wire transfer of immediately available funds.
- **5.3 Calculation of Time**. In this Agreement, a period of days shall be deemed to begin on the first day after the event which began the period and to end at 5:00 p.m. New York, New York time on the last day of the period. If any period of time is to expire hereunder on any day that is not a Business Day, the period shall be deemed to expire at 5:00 p.m. New York, New York time on the next succeeding Business Day.
- **5.4 Schedules and Exhibits.** The Schedules and Exhibits listed below and attached to this Agreement are incorporated herein by reference and deemed to be part of this Agreement.

### **Schedules**

1.0	-	Definitions and Interpretation
2.1(11)	-	Financial Statements
2.1(13)	-	Seismic Data
2.1(15)	-	Licences and Compliance with Applicable Law
2.1(16)	-	Legal Proceedings and Orders
2.1(17)	-	Contracts
2.1(18)	-	Intellectual Property
2.1(19)	-	Environmental Matters
2.1(20)	-	Employees and Employee Benefits
2.1(28)	-	Insurance

4.10 - Corporation Liabilities

A/R Schedule Pre-Closing A/R

**Exhibits** 

Exhibit A Transition Services Agreement

Exhibit B Escrow Agreement

Exhibit C Sample Calculation

- **5.5 Expenses**. Except as otherwise expressly provided herein, each Party shall be responsible for all costs and expenses (including any Taxes imposed on such expenses) incurred by it in connection with the negotiation, preparation, execution, delivery and performance of this Agreement and the transactions contemplated by this Agreement (including the fees and disbursements of legal counsel, bankers, investment bankers, accountants, brokers and other advisers).
- 5.6 Public Announcements. The Parties agree that no public release or announcement related to the transactions contemplated by this Agreement will be issued or made by or on behalf of Purchaser or Corporation, on the one hand, without the prior consent of the Vendor, or by or on behalf of the Vendor, on the other hand, without the prior consent of Purchaser, except as may be required by Applicable Law or regulation or by obligations pursuant to any listing agreement with any national securities exchange or stock market, in which case such Person required to make the release or announcement will allow Vendor or Purchaser, as applicable. reasonable time to comment on such release or announcement in advance of such issuance; provided that in any event the first public announcement with respect to the transactions contemplated by this Agreement shall be mutually acceptable to both Parties. The Parties further acknowledge and agree that the Vendor may disclose such terms and the existence of this Agreement and the transactions contemplated by this Agreement (i) to their Affiliates in order that such Persons may provide information about the subject matter of this Agreement and the transactions contemplated by this Agreement to their respective limited partners and prospective limited partners in connection with their fundraising and reporting activities and (ii) to third parties that are authorized under Section 1(d) of the Transition Services Agreement.

#### 5.7 Notices.

- (1) Mode of Giving Notice. Except as otherwise expressly provided herein, all notices, demands and other communications to be given or delivered under or by reason of the provisions of this Agreement will be in writing and will be deemed to have been given (i) when personally delivered, (ii) when transmitted via facsimile device or by electronic mail (unless if transmitted after 5:00 p.m. Central time or other than on a Business Day, then on the next Business Day), (iii) the day following the day on which the same has been delivered prepaid to a reputable national overnight air courier service or (iv) the third (3rd) Business Day following the day on which the same is sent by certified or registered mail, postage prepaid, in each case, to the respective Party at the number, electronic mail address or street address, as applicable, set forth below, or at such other number, electronic mail address or street address as such Party may specify by written notice to the other Party.
  - (a) if to the Vendor, to:

Seitel Canada Holdings, Inc. 10811 South Westview Circle Drive Bldg. C, #100 Houston, TX 77043

Attn: Marcia H. Kendrick

Email: [email address redacted – confidential information]

with a copy (which shall not constitute notice) to:

Kirkland & Ellis LLP 200 Clarendon Street Boston, MA 02116

Attention: Neal J. Reenan, P.C. and Amanda Border

Email: [email address redacted - confidential information]

[email address redacted – confidential information]

Kirkland & Ellis LLP 300 North LaSalle Street Chicago, IL 60654 Attention: Steve Toth

Email: [email address redacted – confidential information]

(b) if to the Purchaser, to:

Pulse Seismic Inc.

Suite 2700, 421 – 7<sup>th</sup> Avenue S.W.

Calgary, AB T2P 4K9

Attention: Neal Coleman and Pamela Wicks

Email: **[email address redacted – confidential information]** 

[email address redacted – confidential information]

with a copy (which shall not constitute notice) to:

Burgess Energy Advisors 200, 117 – 8<sup>th</sup> Avenue S.W. Calgary, AB T2P 1B4

Attention: Patrick W. Burgess

Email: [email address redacted – confidential information]

- (2) Change of Address. Any Party may from time to time change its address under this Section 5.7 by notice to the other Party given in the manner provided by this Section 5.7.
- **5.8 Time of Essence**. Time shall be of the essence of this Agreement in all respects, subject to Section 1(c) of the Transition Services Agreement.
- **5.9 Further Assurances**. Each Party shall from time to time promptly execute and deliver or cause to be executed and delivered all such further documents and instruments and shall do or cause to be done all such further acts and things in connection with this Agreement that the other Party may reasonably require as being necessary or desirable in order to effectively carry out or better evidence or perfect the full intent and meaning of this Agreement or any provision hereof.

- **5.10 Entire Agreement**. This Agreement, the Transition Services Agreement, the Confidentiality Agreement and the Escrow Agreement constitute the entire agreement between the Parties pertaining to the subject matter of this Agreement and supersedes all prior agreements, understandings, negotiations and discussions, whether oral or written, (including that letter of intent between the Parties dated November 20, 2018). There are no conditions, representations, warranties, obligations or other agreements between the Parties in connection with the subject matter of this Agreement (whether oral or written, express or implied, statutory or otherwise) except as explicitly set out in this Agreement, the Transition Services Agreement, the Confidentiality Agreement or the Escrow Agreement.
- **5.11** Amendment. No amendment of this Agreement shall be effective unless made in writing and signed by the Parties.
- **5.12 Waiver**. A waiver of any default, breach or non-compliance under this Agreement shall not be effective unless in writing and signed by the Party to be bound by the waiver, and then only in the specific instance and for the specific purpose for which it has been given. No waiver shall be inferred from or implied by any failure to act or delay in acting by a Party in respect of any default, breach or non-observance or by anything done or omitted to be done by the other Party. The waiver by a Party of any default, breach or non-compliance under this Agreement will not operate as a waiver of that Party's rights under this Agreement in respect of any continuing or subsequent default, breach or non-observance (whether of the same or any other nature).
- **5.13 Severability**. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction will, as to that jurisdiction, be ineffective to the extent of such prohibition or unenforceability and will be severed from the balance of this Agreement, all without affecting the remaining provisions of this Agreement or affecting the validity or enforceability of such provision in any other jurisdiction.
- Governing Law; Jurisdiction. All issues and questions concerning the construction, validity, interpretation and enforceability of this Agreement, and all claims and disputes arising hereunder or in connection herewith, whether purporting to sound in contract or tort, or at law or in equity, shall be governed by, and construed in accordance with, the laws of the State of Delaware, without giving effect to any choice of Applicable Law or conflict of Applicable Law rules or provisions (whether of the State of Delaware or any other jurisdiction) that would cause the application of the Applicable Laws of any jurisdiction other than the State of Delaware. The Parties hereby agree and consent to be subject to the exclusive jurisdiction of the Court of Chancery of the State of Delaware or, to the extent such court declines jurisdiction, first to any federal court, or second, to any state court, each located in Wilmington, Delaware, and hereby waive the right to assert the lack of personal or subject matter jurisdiction or improper venue in connection with any such suit, action or other proceeding. In furtherance of the foregoing, each of the Parties (i) waives the defense of inconvenient forum, (ii) agrees not to commence any suit, action or other proceeding arising out of this Agreement or any transactions contemplated hereby other than in any such court, and (iii) agrees that a final judgment in any such suit, action or other proceeding shall be conclusive and may be enforced in other jurisdictions by suit or judgment or in any other manner provided by Applicable Law. EACH PARTY HEREBY IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY LITIGATION, ACTION, PROCEEDING, CROSS-CLAIM, OR COUNTERCLAIM IN ANY COURT (WHETHER BASED ON CONTRACT, TORT, OR OTHERWISE) ARISING OUT OF, RELATING TO OR IN CONNECTION WITH (a) THIS AGREEMENT OR THE VALIDITY, PERFORMANCE, INTERPRETATION, COLLECTION OR ENFORCEMENT HEREOF OR (b) THE ACTIONS OF

SUCH PARTY IN THE NEGOTIATION, AUTHORIZATION, EXECUTION, DELIVERY, ADMINISTRATION, PERFORMANCE OR ENFORCEMENT HEREOF.

- **5.15** Successors and Assigns; Assignment. This Agreement shall inure to the benefit of, and be binding on, the Parties and their respective successors and permitted assigns. Neither Party may assign or transfer, whether absolutely, by way of security or otherwise, all or any part of its respective rights or obligations under this Agreement without the prior written consent of the other Party, other than to an Affiliate of such first Party.
- **5.16 Third Party Beneficiaries**. This Agreement is for the sole benefit of the Parties, and except as specifically provided for in Section 4.3, and nothing in this Agreement, express or implied, is intended to or shall confer upon any other Person any legal or equitable right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- **5.17 Counterparts.** This Agreement may be executed in counterparts, each of which shall be deemed to be an original and both of which taken together shall be deemed to constitute one and the same instrument. To evidence its execution of an original counterpart of this Agreement, a Party may send a copy of its original signature on the execution page hereof to the other Party by facsimile, e-mail in pdf format or by other electronic transmission and such transmission shall constitute delivery of an executed copy of this Agreement to the receiving Party.
- **5.18 Specific Performance**. Each of Parties acknowledges and agrees that, subject to Section 1(c) of the Transition Services Agreement, the other Parties would be damaged irreparably in the event any provision of this Agreement is not performed in accordance with its specific terms or is otherwise breached. Accordingly, each of Parties agrees that the other Parties shall be entitled to an injunction or injunctions to prevent breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof in any action instituted in any court in the United States, Canada or in any state or province having jurisdiction over the Parties and the matter in addition to any other remedy to which they may be entitled pursuant hereto. Each of the Parties further agree not to raise any objections to the availability of the equitable remedy of specific performance or injunction to prevent breaches or threatened breaches of, or to enforce compliance with, the covenants and obligations of the Parties under this Agreement. Each of the Parties hereby waive any requirement under any Applicable Law to post a bond or other security as a prerequisite to obtaining equitable relief of any kind.
- **5.19 Non-Recourse**. This Agreement may only be enforced against, and any claim or suit based on, arising out of or related to this Agreement or the negotiation, execution or performance of this Agreement may only be brought against, the named Parties to this Agreement and then only with respect to the specific obligations set forth herein with respect to the named Parties to this Agreement. No Person who is not a named party to this Agreement, including any past, present or future director, manager, officer, employee, incorporator, member, partner, direct or indirect equityholder, Affiliate or Representative of the Corporation, the Vendor or any of their respective Affiliates will have or be subject to any liability or indemnification obligation (whether in contract or in tort) to the Purchaser or any other Person resulting from (nor will the Purchaser have any claim with respect to) (i) the distribution to the Purchaser or its Representatives or the Purchaser's or its Representatives' use of or reliance on any information, documents, projections, forecasts or other material made available to Purchaser in certain "data rooms," confidential information memoranda or management presentations in expectation of, or in connection with, the transactions contemplated by this

Agreement or (ii) any claim based on, in respect of or by reason of the sale and purchase of the Corporation, including any alleged non-disclosure or misrepresentations made by any such Persons or other Persons, in each case, regardless of the legal theory under which such liability or obligation may be sought to be imposed, whether sounding in contract or tort, or whether at law or in equity, or otherwise; and each Party waives and releases all such liabilities and obligations against any such Persons. No Person who is not a named party to this Agreement, including any past, present or future director, manager, officer, employee, incorporator, member, lender, partner, direct or indirect equityholder, Affiliate or Representative of the Purchaser or any of their respective Affiliates will have or be subject to any liability or indemnification obligation (whether in contract or in tort) to the Vendor or any other Person resulting from (nor will the Vendor have any claim with respect to) any claim based on, in respect of or by reason of the sale and purchase of the Corporation, or the financing thereof, including any alleged non-disclosure or misrepresentations made by any such Persons or other Persons, in each case, regardless of the legal theory under which such liability or obligation may be sought to be imposed, whether sounding in contract or tort, or whether at law or in equity, or otherwise; and each Party waives and releases all such liabilities and obligations against any such Persons.

- **5.20 Non-Competition**. For a period of three (3) years from Closing, the Vendor shall not, act directly or indirectly, alone or as a partner, consultant, agent, contractor, joint venture or shareholder of any entity that:
- (1) is in competition with the Business within the Provinces of Alberta, British Columbia or Saskatchewan, in Canada; or
- (2) intends to develop, license, distribute, market, lease or sell products or services within the Provinces of Alberta, British Columbia or Saskatchewan, in Canada which are in direct competition with the products or services being developed, licensed, distributed, marketed, provided or sold by the Purchaser.

[Signature page follows.]

IN WITNESS WHEREOF the Parties have executed this Agreement as of the date first above written.

# SEITEL CANADA HOLDINGS, INC.

By: "Richard C. Kelvin"

Name: Richard C. Kelvin

Title: President

# **PULSE SEISMIC INC.**

By: "Neal Coleman"

Name: Neal Coleman

Title: President and Chief Executive Officer

# SCHEDULE 1.0 DEFINITIONS

### Definitions.

- "2017 Audit" has the meaning set out in Section 4.2.
- "2018 Audit" has the meaning set out in Section 4.2.
- "A/R Cash" has the meaning set out in Section 1.11(1).
- "A/R Cash Payment" has the meaning set out in Section 1.11(1).
- "A/R Cash Statement" has the meaning set out in Section 1.11(2).
- "A/R Period" has the meaning set out in Section 1.11(2).
- "Accounting Records" means all of the Corporation's books of account, accounting records and other financial information (whether in written, printed, electronic or computer printout form, or stored electronically, digitally or on computer related media).
- "Adjustment Date" means the fifth Business Day after the Closing Working Capital is finally determined in accordance with Section 1.8(1) or Section 1.8(2), as the case may be.
- "Affiliate" means, with respect to any Person, any other Person who directly or indirectly controls, is controlled by, or is under direct or indirect common control with, such Person, and includes any Person in like relation to an Affiliate. A Person shall be deemed to "control" another Person if such Person possesses, directly or indirectly, the power to direct or cause the direction of the management and policies of such other Person, whether through the ownership of voting securities, by contract or otherwise; and the term "controlled" shall have a similar meaning.
- "Agreement" means the Share Purchase Agreement to which this Schedule 1.0 is attached, together with all the Exhibits and the Schedules attached thereto.
- "Alternative Arrangements" has the meaning set out in Section 3.6.
- "Annual Financial Statements" means the annual unaudited financial statements of the Corporation for the fiscal years ended December 31, 2016 and December 31, 2017, true and complete copies of which are attached as Schedule 2.1(11).
- "Applicable Law" means, with respect to any Person, property, transaction, event or other matter, (i) any foreign or domestic constitution, treaty, law, statute, regulation, code, ordinance, principle of common law or equity, rule, municipal by-law, Order or other requirement having the force of law, (ii) any policy, practice, protocol, standard or guideline of any Canadian Governmental Authority which, although not necessarily having the force of law, is regarded by such Governmental Authority as requiring compliance as if it had the force of law, in each case relating or applicable to such Person, property, transaction, event or other matter and also includes, where appropriate, any interpretation of Applicable Law (or any part thereof) by any Person having jurisdiction over it, or charged with its administration or interpretation.

- "Auditor" has the meaning set out in Section 4.2.
- "Books and Records" means the Accounting Records and all other information in any form to the extent relating to the Business or the Corporation, including sales and purchase records, list of customers, lists of potential customers, credit and pricing information, personnel and payroll records, tax records, business reports, plans and projections, production reports and records, inventory reports and records, business and consulting reports, marketing and advertising materials, and research and development reports (whether in written, printed, electronic or computer printout form, or stored electronically, digitally or on computer related media).
- "Business" means the business carried on by the Corporation consisting of the licensing, leasing, acquisition and storage of Seismic Data.
- "Business Day" means any day except Saturday, Sunday or any day on which banks are generally not open for business in the City of New York, New York.
- "Canadian Dollars" means the lawful currency of Canada.
- "Cash Resales" means new contracts pursuant to which the Corporation licenses Seismic Data and that are payable in cash.
- "Cash Resale Amount" has the meaning set forth in Section 1.12(1).
- "Cash Resale Objection Notice" has the meaning set forth in Section 1.12(3).
- "Cash Resale Period" has the meaning set forth in Section 1.12(2).
- "Cash Resale Statement" has the meaning set forth in Section 1.12(2).
- "Cash Resale Term" has the meaning set forth in Section 1.12(1).
- "Claim Notice" has the meaning set out in Section 3.4.
- "Closing" means the completion of the purchase and sale of the Shares in accordance with the provisions of this Agreement.
- "Closing Balance Sheet" means the balance sheet prepared in accordance with the provisions of Section 1.7(1)(a).
- "Closing Date" means the date of this Agreement, unless agreed otherwise by the Parties.
- "Closing Document" means any agreement, certificate or other instrument to be executed or delivered at the Closing as contemplated by this Agreement.
- "Closing Time" has the meaning set out in Section 1.2.
- "Closing Working Capital" means the working capital of the Corporation calculated in accordance with the provisions of Section 1.7(1)(b).

- "Confidentiality Agreement" means that certain Confidentiality Agreement, dated March 26, 2018, by and between Pulse Seismic Inc. and Seitel, Inc., as amended by that certain Addendum and Amendment to Confidentiality Agreement, dated as of September, 2018, by and between Seitel, Inc. and Pulse Seismic Inc.
- "Contract" means any written contract, agreement, instrument or other legally binding commitment or arrangement to which the Corporation is a party or under which it has rights or obligations, included those listed or described in any Schedule.
- "Corporation" means Seitel Canada Ltd., a corporation incorporated under the laws of Alberta.
- "Corporation Intellectual Property" has the meaning set out in Section 2.1(18)(a).
- "Corporation Liabilities" means the liabilities of the Corporation specifically set forth on Schedule 4.10 hereof, calculated in accordance with Schedule 4.10 on the 12 month anniversary of the date hereof.
- "Current Assets" means, for the purposes of this Agreement, current assets specifically set forth in the Sample Calculation.
- "Current Liabilities" means, for the purposes of this Agreement, current liabilities specifically set forth in the Sample Calculation; <u>provided</u>, for the avoidance of doubt, in no event shall Current Liabilities include any amounts included in Corporation Liabilities.
- "Damages" has the meaning set out in Section 3.2.
- "Dataroom" has the meaning set out in Section 2.4.
- "Direct Claim" has the meaning set out in Section 3.4.
- "**Employee**" means an individual who is employed by the Corporation, whether on a full-time or part-time basis.
- "Employee Plans" has the meaning set out in Section 2.1(20)(e).
- "Environmental Law" means Applicable Law in respect of the protection of the natural environment or any species or organisms that make use of it, public or occupational health or safety, or the manufacture, importation, handling, transportation, storage, disposal and treatment of Hazardous Substances.
- "Equity Interest" means any capital stock, partnership or limited liability company interest (as applicable), other equity (including profit participation features, stock or equity appreciation rights, "phantom" stock rights or other similar rights) or voting interest or proxy or any security or evidence of indebtedness directly or indirectly convertible into or exchangeable or exercisable for any capital stock, partnership or limited liability company interest, or any warrant or option or other rights, directly or indirectly, to subscribe for (including any preemptive rights) or otherwise acquire any of the foregoing.
- "Escrow Account" has the meaning set out in the preamble to this Agreement.

- "Escrow Agent" mean Blake, Cassels & Graydon LLP.
- "Escrow Agreement" has the meaning set out in the preamble to this Agreement.
- "Escrow Amount" has the meaning set out in Section 1.10(2).
- "Estimated Purchase Price" has the meaning calculated by the provisions of Section 1.6.
- "Estimated Working Capital" has the meaning set out in Section 1.6(1).
- "Expected Working Capital" has the meaning calculated by the provisions of Section 1.6(2).
- "Express Representations" has the meaning set out in Section 2.4.
- "Final Determination" means a determination made by a Governmental Authority (including pursuant to a settlement) where all rights to object to or appeal from the determination (including any right to obtain relief under a competent authority or similar process) have been exhausted or have expired.
- "Financial Statements" means, collectively, the Annual Financial Statements and the Interim Financial Statements.
- "Fraud" means, with respect to (i) the Vendor, the actual and intentional misrepresentation by the Vendor in the making by the Vendor to Purchaser of the representations and warranties set forth in Section 2.1 or (ii) Purchaser, the actual and intentional misrepresentation by such Person in making by such Person to the Corporation or Vendor of the representations and warranties set forth in Section 2.2, in each case of clause (i) or (ii), that constitutes common law fraud under Delaware law (and does not include any fraud claim based on constructive knowledge, negligent misrepresentation, recklessness or a similar theory).
- "GAAP" or "generally accepted accounting principles" means Canadian private company generally accepted accounting principles, as in effect from time to time.

## "Governmental Authority" means:

- (a) any domestic or foreign government, whether national, federal, provincial, state, territorial, municipal or local (whether administrative, legislative, executive or otherwise);
- (b) any agency, authority, ministry, department, regulatory body, court, central bank, bureau, board or other instrumentality having legislative, judicial, taxing, regulatory, prosecutorial or administrative powers or functions of, or pertaining to, government;
- (c) any court, tribunal, commission, individual, arbitrator, arbitration panel or other body having adjudicative, regulatory, judicial, quasi-judicial, administrative or similar functions; and

- (d) any other body or entity created under the authority of or otherwise subject to the jurisdiction of any of the foregoing, including any stock or other securities exchange or professional association.
- "Hazardous Substance" means any solid, liquid, gas, odour, heat, sound, vibration, radiation or combination of them that may impair the natural environment, injure or damage property or plant or animal life or harm or impair the health of any individual and includes any contaminant, waste, or substance or material defined, prohibited, regulated or reportable pursuant to any Environmental Law.
- "Indemnified Party" means a Person whom the Vendor or the Purchaser, as the case may be, is required to indemnify under Article 3.
- "Indemnifying Party" means, in relation to an Indemnified Party, the Party that is required to indemnify such Indemnified Party under Article 3.
- "Independent Accountant" has the meaning set out in Section 1.8(2).
- "Intellectual Property" means all of the following: (i) patents, patent applications and patent disclosures; (ii) trademarks, service marks, trade dress, corporate names and Internet domain names, together with all goodwill associated with each of the foregoing; (iii) copyrights; (iv) registrations and applications for any of the foregoing; (v) trade secrets; (vi) computer software; (vii) drawings, schematics and other technical plans; and (viii) all other intellectual property.
- "Interim Financial Statements" means the unaudited financial statements of the Corporation for the 9 month period ended September 30, 2018, true and complete copies of which are attached as Schedule 2.1(11).
- "ITA" means the Income Tax Act (Canada) and the regulations thereunder.
- "Joint Venture Agreements" means all joint venture and other ownership agreements pertaining to the Seismic Data set forth on Schedule 2.1(13)(a).
- "Leased Real Property" has the meaning set out in Section 2.1(14).
- "Legal Proceeding" means any litigation, action, application, suit, investigation, hearing, claim, complaint, deemed complaint, grievance, civil, administrative, regulatory or criminal, arbitration proceeding or other similar proceeding, before or by any Governmental Authority, and includes any appeal or review thereof and any application for leave for appeal or review.
- "Liabilities" has the meaning set out in Section 2.1(11)(b).
- "License" means any license, permit, approval, authorization, certificate, directive, order variance, registration (excluding as to Intellectual Property), right, privilege, concession or franchise, in each case, issued, granted, conferred or otherwise created by any Governmental Authority.
- "Liens" means liens, security interests, charges or encumbrances. Liens do not include non-exclusive licenses of Intellectual Property.

"Material Adverse Change" means any materially adverse change to the financial condition, business or results of operations of the Corporation, but shall exclude any prospects and shall also exclude any change resulting or arising from: (i) any change in any Applicable Law; (ii) any change in interest rates, credit markets, currency exchange rates or general economic conditions (including changes in the price of gas, oil or other natural resources or chemicals); (iii) any change that is generally applicable to the industries in which the Corporation operates; (iv) the entry into this Agreement or the announcement or consummation of the transactions contemplated by this Agreement and the other agreements referenced herein; (v) any action taken by or at the request of the Purchaser or any of its Affiliates; (vi) any omission to act or action taken with the consent of the Purchaser (including those omissions to act or actions taken which are permitted by this Agreement); (vii) any national or international political event or occurrence, including acts of war or terrorism; (viii) any natural or manmade disasters, hurricanes, floods, tornados, pandemics, tsunamis, earthquakes or other acts of God; (ix) any actions required in order to obtain any waiver or consent from any Person or Governmental Authority in connection with the transactions contemplated by this Agreement and the other agreements referenced herein; (x) the availability or cost of equity, debt or other financing to Purchaser; (xi) any item or matter set forth on the Schedules attached hereto; or (xii) any failure by the Corporation to meet any projections, forecasts or estimates of revenue or earnings (it being understood that this clause (xii) shall not prevent a determination that any change underlying such failure to meet projections, forecasts or estimates has resulted in a Material Adverse Change (to the extent the effect(s) of such change is not otherwise excluded from this definition of Material Adverse Change)); provided, that, in the case of the foregoing clauses (i), (ii), (iii), (vii) and (viii), if such effect disproportionately affects the Corporation as compared to other Persons or businesses that operate in the industry in which the Corporation operates, then the disproportionate aspect of such effect may be taken into account in determining whether a Material Adverse Change has occurred or will occur.

"Notice of Acceptance" has the meaning set out in Section 1.8(1).

"Notice of Objection" has the meaning set out in Section 1.8(1).

"Order" means any order, directive, judgment, decree, injunction, decision, ruling, award or writ of any Governmental Authority.

"Party" means a party to this Agreement and any reference to a Party includes its successors and permitted assigns and "Parties" means every Party.

"Per Claim Threshold" has the meaning set out in Section 3.5.

"Permitted Liens" means (i) Liens securing liabilities which are reflected or reserved against in the Interim Financial Statements to the extent so reflected or reserved; (ii) Liens for Taxes not yet delinquent or which are being contested in good faith; (iii) landlord's, mechanic's, materialmen's, and other similar statutory Liens arising or incurred in the ordinary course of business for amounts not yet due and payable or which are being contested in good faith; (iv) purchase money Liens and Liens securing rental payments under capital lease arrangements; (v) zoning, building codes and other land use laws regulating the use or occupancy of real property or the activities conducted thereon which are imposed by any Governmental Authority which are not violated by the current use or occupancy of such real property or the operation of the

business or any violation of which would not reasonably be expected to be material to the Corporation, taken as a whole; (vi) easements, rights, covenants, conditions and restrictions of record; (vii) Liens arising in the ordinary course of business (including customer return rights, warranty claims, rebates, refunds and other discounts to customers) and not incurred in connection with the borrowing of money; (viii) Liens in the form of security deposits or otherwise collateralized with letters of credit or similar instruments; (ix) Liens that will be terminated in connection with or prior to the Closing; and (x) Liens that would not reasonably be expected to be material to the Corporation, taken as a whole.

- "Person" is to be broadly interpreted and includes an individual, a corporation, a partnership, a trust, an unincorporated organization, a Governmental Authority, and the executors, administrators or other legal Representatives of an individual in such capacity.
- "Post-Closing Adjustment Amount" has the meaning set out in Section 1.9(1).
- "Pre-Closing A/R" has the meaning set out in Section 1.11(1).
- "Pre-Closing A/R Cash Amount" has the meaning set out in Section 1.11(1).
- "Pre-Closing Tax Period" means a taxation year that ends on or before the date hereof.
- "Prime Rate" for any day means the rate of interest expressed as a rate per annum that the Toronto-Dominion Bank establishes as the reference rate of interest that it will charge on that day for Canadian dollar demand loans to its customers in Canada and which it refers to as its prime rate.
- "Purchase Price" has the meaning set out in Section 1.5.
- "Purchaser" has the meaning set out in the preamble to the Agreement.
- "Purchaser's Indemnified Parties" means the Purchaser, the Purchaser's Affiliates, the Corporation and their respective Representatives.
- "Regulatory Approval" means any approval, consent, ruling, authorization, notice, permit, waiver or acknowledgement that may be required from any Person pursuant to Applicable Law or under the terms of any licence or the conditions of any Order (i) in connection with the transactions contemplated by this Agreement, (ii) to permit the Corporation to carry on its business after Closing, or (iii) which is otherwise necessary to permit the Parties to perform their obligations under this Agreement.
- "Representative" when used with respect to a Person means each director, officer, employee, consultant, financial adviser, legal counsel, accountant and other agent, adviser or representative of that Person.
- "Review Period" means 30 days following the receipt of the Closing Balance Sheet and the calculation of the Closing Working Capital by the Vendor.

- "Sample Calculation" means the sample calculation of Net Working Capital as of January 15, 2019 set forth in Exhibit C.
- "Schedules" has the meaning set out in Section 1(a)(11) of Schedule 2.0.
- "Seismic Data" means all of the field data set forth on Schedule 2.1(13)(a) that is stored on the TSM Database and originates within the Provinces of Alberta, British Columbia or Saskatchewan, in Canada.
- "Seismic Data Licenses" has the meaning set forth in Section 2.1(13)(b).
- "Service Fees" has the meaning set out in Section 1.11(1).
- "Shares" has the meaning set out in Section 2.1(3).
- "Straddle Period" means the taxation year or fiscal period that includes, but does not begin or end on, the date hereof.
- "Stub Period Returns" has the meaning set out in Section 4.1.
- "Tax Returns" means all returns, reports, declarations, elections, notices, filings, information returns, and statements in respect of Taxes that are filed or required to be filed with any applicable Governmental Authority, including all amendments, schedules, attachments or supplements thereto and whether in tangible or electronic form.
- "Taxes" means, with respect to any Person, all supranational, national, federal, provincial, state, local or other taxes, including income taxes, branch taxes, profits taxes, capital gains taxes, gross receipts taxes, windfall profits taxes, value added taxes, severance taxes, ad valorem taxes, property taxes, capital taxes, net worth taxes, sales taxes, use taxes, licence taxes, excise taxes, franchise taxes, environmental taxes, transfer taxes, withholding or similar taxes, payroll taxes, employment taxes, employer health taxes, government pension plan premiums and contributions, social security premiums, workers' compensation premiums, employment/unemployment insurance or compensation premiums and contributions, stamp taxes, occupation taxes, premium taxes, alternative or add-on minimum taxes, goods and services tax, harmonized sales tax, customs duties or other taxes of any kind whatsoever imposed or charged by any Governmental Authority and any instalments in respect thereof, interest, penalties, or additions with respect thereto and any interest in respect of such additions or penalties, and whether disputed or not, and "Tax" means any one of such Taxes.
- "Third Party Claim" has the meaning set out in Section 3.4.
- "Threatened", when used in relation to a Legal Proceeding or other matter, means that a written demand or statement has been made or a written notice has been given that a Legal Proceeding or other matter is being asserted, commenced, taken or otherwise pursued.
- "Threshold" has the meaning set forth in Section 3.5(b).
- "Transaction Accounting Principles" has the meaning set forth in Section 1.7(1)(a).

- "Transition Services Agreement" means the Transition Services Agreement attached hereto as Exhibit A.
- "TSM Database" means the data structure used by the Business immediately prior to Closing in connection with the storage of its Seismic Data.
- "Vendor" has the meaning set out in the preamble to the Agreement.
- "Vendor's Indemnified Parties" means the Vendor, the Vendor's Affiliates and their Representatives.
- **"Working Capital"** means, at any time, the Corporation's Current Assets less its Current Liabilities at that time, calculated in accordance with the Sample Calculation and the Transaction Accounting Principles.

# SCHEDULE 2.0 Additional Rules of Interpretation.

- (1) Gender and Number. In this Agreement, unless the context requires otherwise, words in one gender include all genders and words in the singular include the plural and vice versa.
- (2) Headings and Table of Contents. The inclusion in this Agreement of headings of Articles and Sections and the provision of a table of contents are for convenience of reference only and are not intended to be full or precise descriptions of the text to which they refer.
- (3) Section References. Unless the context requires otherwise, references in this Agreement to Articles, Sections, Schedules or Exhibits are to Articles or Sections of this Agreement, and Schedules or Exhibits to this Agreement.
- (4) Accounting Terms. Accounting terms which are not otherwise defined in this Agreement have the meanings given to them under GAAP. To the extent that the definition of an accounting term defined in this Agreement is inconsistent with the meaning of such term under GAAP, the definition set forth in this Agreement will control.
- (5) Words of Inclusion. Wherever the words "include", "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation" and the words following "include", "includes" or "including" shall not be considered to set forth an exhaustive list.
- (6) References to this Agreement. The words "hereof", "herein", "hereto", "hereunder", "hereby" and similar expressions shall be construed as referring to this Agreement in its entirety and not to any particular Section or portion of it.
- (7) Statute References. Unless otherwise indicated, all references in this Agreement to any statute include the regulations thereunder, in each case as amended, re-enacted, consolidated or replaced from time to time and in the case of any such amendment, re-enactment, consolidation or replacement, reference herein to a particular provision shall be read as referring to such amended, re-enacted, consolidated or replaced provision and also include, unless the context otherwise requires, all applicable guidelines, bulletins or policies made in connection therewith; provided that, for the purposes of the representations and warranties set forth herein, with respect to any violation of or non-compliance with, or alleged violation of or non-compliance, with any code section or Applicable Law, the reference to such code section or Applicable Law means such code section or Applicable Law as in effect at the time of such violation or non-compliance or alleged violation or non-compliance.
- (8) Document References. All references herein to any agreement (including this Agreement), document or instrument mean such agreement, document or instrument as amended, supplemented, modified, varied, restated or replaced from time to time in accordance with the terms thereof and, unless otherwise specified therein, includes all schedules and exhibits attached thereto.
- (9) Mandatory; Permissive. The word "will" will be construed to have the same meaning and effect as the word "shall". The words "shall," "will," or "agree(s)" are mandatory, and "may" is permissive.

- (10) Knowledge. Where any representation, warranty or other statement in this Agreement is expressed to be made by the Vendor to its knowledge or is otherwise expressed to be limited in scope to facts or matters known to the Vendor or the Corporation or of which the Vendor or the Corporation is aware, it shall mean the actual knowledge of Robert Monson, Marcia Kendrick or Richard Kelvin. Where any representation, warranty or other statement in this Agreement is expressed to be made by the Purchaser to its knowledge or is otherwise expressed to be limited in scope to facts or matters known to the Purchaser or of which the Purchaser is aware, it shall mean the actual knowledge of Neal Coleman or Pamela Wicks.
- Schedules. The information set forth in each Section or sub-Section of the disclosures in the schedules referred to in this Agreement (the "Schedules") are to be taken as relating to the representations and warranties of the Corporation and the Vendor as a whole, notwithstanding the fact that the Schedules are arranged by sections corresponding to the sections in this Agreement or that a particular section of this Agreement makes reference to a specific section of the Schedules and notwithstanding that a particular representation and warranty may not make a reference to the Schedules. The inclusion of information in the Schedules will not be construed as, and will not constitute, an admission or agreement that a violation, right of termination, default, liability or other obligation of any kind exists with respect to any item, nor will it be construed as or constitute an admission or agreement that such information is material to the Corporation, the Vendor or the Purchaser. In addition, matters reflected in the Schedules are not necessarily limited to matters required by this Agreement to be reflected in the Schedules. Such additional matters are set forth for informational purposes only and do not necessarily include other matters of a similar nature. Neither the specification of any dollar amount in any representation, warranty or covenant contained in this Agreement nor the inclusion of any specific item in the Schedules is intended to imply that such amount, or higher or lower amounts, or the item so included or other items, are or are not material, and no Person will use the fact of the setting forth of any such amount or the inclusion of any such item in any dispute or controversy between the Parties as to whether any obligation, item or matter not described in this Agreement or included in the Schedules is or is not material for purposes of this Agreement. Further, neither the specification of any item or matter in any representation, warranty or covenant contained in this Agreement nor the inclusion of any specific item in the Schedules is intended to imply that such item or matter, or other items or matters, are or are not in the ordinary course of business, and no Person will use the fact of setting forth or the inclusion of any such item or matter in any dispute or controversy between the Parties as to whether any obligation, item or matter not described in this Agreement or included in the Schedules is or is not in the ordinary course of business for purposes of this Agreement.

# Exhibit A

Transition Services Agreement
[Attached]

#### TRANSITION SERVICES AGREEMENT

THIS TRANSITION SERVICES AGREEMENT (this "Agreement") is made and entered into as of January 15, 2019 ("Effective Date"), by and between Seitel, Inc., a Delaware corporation (the "Supplier Party"), on the one hand, and Seitel Canada Ltd., a corporation incorporated under the laws of Alberta (the "Purchaser Party"), on the other hand. The Supplier Party and the Purchaser Party are each referred to herein as a "Party" and collectively as the "Parties". Capitalized terms not otherwise defined herein shall have the meanings set forth in the Purchase Agreement (as defined below).

**WHEREAS,** Seitel Canada Holdings, Inc., a Delaware corporation ("Seller") and Pulse Seismic Inc., a Canadian corporation ("Purchaser") are parties to that certain Share Purchase Agreement, dated as of January 15, 2019, (the "Purchase Agreement"), pursuant to which, among other things, Seller is selling to the Purchaser the Shares; and

**WHEREAS,** the Supplier Party has agreed to provide, or cause its Affiliates to provide, to the Purchaser Party, from and after the Effective Date, certain transition services on the terms and conditions set forth herein.

**NOW, THEREFORE,** in consideration of the foregoing and the mutual covenants, conditions and provisions contained herein, the Supplier Party and the Purchaser Party hereby agree as follows.

## 1. <u>Transition Services to be Provided by the Supplier Party.</u>

- (a) Transition Services. From the Effective Date through the Termination Date (as defined in Section 4(c) below), the Supplier Party shall, and shall cause its Affiliates to, provide to the Purchaser Party, on the terms set forth herein, solely the services specifically listed on Schedule A attached hereto (the "Transition Services"); provided however, that the Transition Services shall not include any of the following: (i) Purchaser Party having direct access to or use of Supplier's equipment, software, systems, databases, communications networks, or facilities, including any of the forgoing that are used by Supplier but owned by a third party (together, "Supplier Party Systems") that would require Supplier Party to obtain a license or any third party's consent on behalf of Purchaser Party to use such Supplier Party Systems; (ii) data migration services; or (iii) software development, configuration or customization services (individually and collectively, "Excluded Services").
- (b) <u>Standard of Performance</u>. The Supplier Party shall, and shall cause its Affiliates to, provide to the Purchaser Party the Transition Services in substantially the same manner, with substantially the same degree of care as the Supplier Party and/or its Affiliates have provided the Transition Services to the Corporation during the twelve (12)-month period immediately prior to the Effective Date.
- (c) <u>Right to Determine Priority</u>. If there is a conflict between the business needs of the Supplier Party or its Affiliates and the provision of any Transition Service to the Purchaser Party, the Supplier Party shall have the right, in its sole and absolute discretion, to establish reasonable priorities, including but not limited to particular times and under particular circumstances, as between the business of and services provided to the Supplier Party and its Affiliates, on the one hand, and the Transition Services provided to Purchaser Party pursuant to this Agreement, on the other hand. Nothing in this Agreement shall require the Supplier Party to favor the Purchaser Party over

the business of the Supplier Party.

### (d) Cooperation; Access.

- (i) During the term of this Agreement, the Parties shall use commercially reasonable efforts to cooperate with each other in all matters relating to the provision and receipt of the Transition Services. Such reasonable cooperation shall include exchanging information, providing physical and electronic access to Purchaser Party's systems required in order for Supplier Party to perform the Transition Services, performing true-ups and adjustments and Purchaser Party shall obtain all consents, licenses, sublicenses or approvals necessary to permit each Party to perform its obligations hereunder, in each case subject to the restrictions of Section 6 hereof. The Parties shall use commercially reasonable efforts to cooperate with each other in determining the extent to which any Tax is due and owing with respect to the Transition Services and in providing and making available any resale certificate, information regarding out-of-state use of materials, services or sale, and other exemption certificates or information reasonably requested by either Party.
- (ii) Purchaser Party acknowledges that Supplier Party's delivery of the Transition Services is dependent on Purchaser Party's cooperation, assistance, and provision of access as reasonably requested by Supplier Party. Purchaser Party further acknowledges that Supplier Party shall not provide any of the Excluded Services and has no duty to cooperation regarding the same.
- (iii) Supplier Party's nonperformance of any of its obligations under this Agreement shall be excused if and to the extent such nonperformance results from Purchaser Party's failure to perform any of its obligations hereunder. Supplier Party and its third party services providers shall not be liable for any impairment of, or failure to provide or make available, any Transition Service to the extent such impairment or failure was the direct result of Purchaser Party's operations or systems or any act or omissions of Purchaser Party (including Supplier Party not receiving the information, materials or access required under this Agreement, either timely or at all), or by its receiving inaccurate or incomplete information on which it reasonably relied from Purchaser Party or any of its Affiliates that is required or reasonably requested regarding such Transition Service. Supplier Party and its third party service providers shall be entitled to rely upon the genuineness, validity or truthfulness of any document, instrument or other writing provided by or on behalf of Purchaser Party or any of its Affiliates in connection with this Agreement.

## (e) Security.

- (i) Supplier Party may in its reasonable discretion suspend the provision of Transition Services (or any part thereof) for reasons of preventative or emergency maintenance with respect to the information technology or computer systems used or relied on by Supplier Party. To the extent practicable, Supplier Party shall use its reasonable efforts to inform Purchaser Party five (5) business days in advance of any such suspension and shall cooperate with Purchaser Party to mitigate any negative effects on Purchaser Party's operations due to such suspension.
- (f) <u>Transitional Nature of Services</u>. Purchaser Party acknowledges and agrees that Supplier Party is providing (or causing to be provided) the Transition Services on a temporary transitional basis to Purchaser Party in order to allow Purchaser Party a period of time to provide or obtain similar services for itself, and that Supplier Party is not a

commercial Supplier Party of such Transition Services. Accordingly, Purchaser Party agrees to, as promptly as reasonably practicable following the Effective Date, transition the Transition Services to its own internal organization or obtain alternate third-party sources to provide the Transition Services and end its use of the Transition Services. For the avoidance of doubt, the Transition Services shall not include, and Supplier Party shall have no obligation under this Agreement to provide (or cooperate or assist with), any of the Excluded Services.

### (g) <u>Intellectual Property Rights</u>.

- (i) <u>Licenses</u>. Purchaser Party, on behalf of itself and its Affiliates, hereby grants to Supplier Party and its authorized third party providers, during the Term, a fully-paid-up, royalty free, non-exclusive, non-transferable license to use any Intellectual Property owned by Purchaser Party or any of its Affiliates that is necessary for Supplier Party and such third party providers to provide the Transition Services, solely to the extent necessary for Supplier Party and such third party providers to provide the Transition Services as required under this Agreement. Nothing in this Agreement shall be deemed to convey to a Party any rights in or to the Intellectual Property of the other Party or its Affiliates other than the rights expressly granted in this <u>Section 1(g)(i)</u>. Each Party shall retain all right, title, and interest in and to its Intellectual Property.
- (ii) Ownership of Work Product. Each Party shall retain all right, title and interest in all materials, products, reports, computer programs (source or object code), documentation, deliverables, and inventions developed, conceived, authored, or prepared by such Party pursuant to this Agreement and all Intellectual Property rights therein.
- (iii) <u>Non-Exclusive Agreement</u>. Nothing in this Agreement shall preclude or restrict Supplier Party from acquiring, marketing, developing, distributing, licensing or using for itself or others (including competitors of Purchaser Party), services, products, materials, or technology that are the same as or similar to those provided to Purchaser Party by Supplier Party pursuant to this Agreement. Furthermore, Supplier Party shall be free to use, without restriction, any knowledge, skills and experience and any ideas, concepts, know-how, and techniques related to the Transition Services that are acquired or used in the course of providing the Transition Services.
- (h) <u>Delegation of Transition Services</u>. Notwithstanding anything to the contrary set forth herein, to the extent any Transition Services were outsourced and provided by a third party during the twelve (12)-month period prior to the Effective Date, the Supplier Party shall be permitted to continue to outsource such Transition Services without the review or approval of the Purchaser Party. Should the Supplier Party wish to outsource or delegate any Transition Services to third parties after the Effective Date, the Supplier Party shall be permitted to do so without the review or approval of the Purchaser Party, if the third party, in the Supplier Party's sole and reasonable judgment, is qualified to provided such Transition Services. The Supplier Party acknowledges that notwithstanding any delegation of their responsibilities under this Agreement, the Supplier Party shall remain responsible for the provision of the Transition Services and the third party's compliance with the standard of performance set forth herein.

#### 2. Pricing and Payment for Transition Services.

(a) Pricing Methodology. The fees for each Transition Service are set

forth on Schedule A and 3D reproduction charges are set forth on Schedule B.

- (b) <u>Invoices</u>. The Supplier Party shall invoice the Purchaser Party monthly in arrears for any fees associated with the Transition Services and, except as otherwise provided in <u>Section 2(d)</u> below, the Purchaser Party shall pay the Supplier Party within fifteen (15) days after receipt of any invoice. Each billing invoice shall set forth the amounts owed for each of the Transition Services, plus applicable taxes. The Purchaser Party shall not offset any amounts payable by it hereunder against any other amounts owed to it by the Supplier Party or any of its Affiliates. If the Purchaser Party fails to pay the full amount within such fifteen (15) period, such failure will be considered a material breach of this Agreement, for which the Supplier Party may, without liability, suspend its or its Affiliates obligations hereunder to provide any and all Transition Services to the Purchaser Party until such time as such invoices have been paid in full.
- (c) <u>Interest Payable on Amounts Past Due</u>. All payments required to be made pursuant to this Agreement shall bear interest from and including the date such payment is due until but excluding the date of payment, at a rate of 5% per annum. Such interest shall be payable at the same time as the payment to which it relates and will be calculated on the basis of a year of 365 days and the actual number of days elapsed, compounded quarterly.
- (d) <u>Taxes</u>. Without limiting any provision hereof, the Party required to pay for any applicable Transition Service hereunder will bear any and all sales, use and transfer taxes and other similar charges (and any related interest and penalties) imposed or payable with respect to the fees, reimbursements and charges payable by it hereunder. If the Purchaser Party or their assignee is required to deduct or withhold any taxes or any amount on account of the Supplier Party's taxes payable to any Canadian governmental authority or agency from any amount payable to the Supplier Party or any of its affiliates in respect of this Agreement, the amount payable to the Supplier Party or applicable affiliates shall be increased as may be necessary so that after making all required deductions (including deductions applicable to additional sums payable under this <u>Section 2(d)</u>), the Supplier Party or its Affiliates receive the amount that they would have received had no such deduction been made.
- (e) <u>No Right to Setoff</u>. There shall be no right of setoff or counterclaim with respect to any claim, debt or obligation, against payments to Supplier Party under this Agreement.

#### 3. Representatives.

(a) <u>Coordinators</u>. The Supplier Party and the Purchaser Party shall each designate, from time to time, a representative to act as the Supplier Party's and the Purchaser Party's respective primary contact persons to coordinate the provision of all of the Transition Services (collectively, the "<u>Primary Coordinators</u>"). Each Primary Coordinator may designate one or more service coordinators for each specific Transition Service (the "<u>Service Coordinators</u>"). Each Party may treat an act of a Primary Coordinator of another Party as being authorized by such other Party without inquiring behind such act or ascertaining whether such Primary Coordinator had authority to so act, and each Party may treat an act of a Service Coordinator as being authorized by such other Party only to the extent such act is directly related to the Transition Service for which such Service Coordinator has been designated; <u>provided</u>, <u>however</u>, that no such Primary Coordinator or Service Coordinator has authority to amend this Agreement. The Supplier Party and the

Purchaser Party shall advise each other promptly (and in any event within seven (7) days) in writing of any change in the Primary Coordinators and any Service Coordinator for a particular Transition Service. The Supplier Party and the Purchaser Party agree that all communications relating to the provision of the Transition Services shall be directed to the Service Coordinators for such Transition Service with copies to the Primary Coordinators.

(b) Disputes. In the event of a dispute between the Parties regarding the terms of this Agreement, the applicable Service Coordinators shall meet in good faith to resolve such dispute within two (2) business days thereof ("Service Coordinators Resolution Period"). If the Parties do not reach a resolution during the Service Coordinators Resolution Period, then the Primary Coordinators shall meet in good faith to resolve such dispute within two (2) business days following the end of the Service Coordinators Resolution Period ("Primary Coordinators Resolution Period"). If the Parties do not reach a resolution during the Primary Coordinators Resolution Period, then Supplier Party shall in its sole, reasonable and good faith discretion, devise an appropriate resolution within one (1) business day following the end of the Primary Coordinators Resolution Period.

#### 4. Term; Termination.

- (a) <u>Term</u>. This Agreement shall become effective upon the Effective Date of this Agreement and shall remain in effect until the expiration or termination of the last Transition Service ("<u>Term</u>"), unless extended in writing signed by the Supplier Party or earlier terminated in accordance with this Agreement. Each Transition Service shall terminate on the earlier of: (i) the expiration date for such Transition Service specified on <u>Schedule A</u>; (ii) the termination of such Transition Service pursuant to <u>Section 4(b)</u>; and (iii) the expiration or termination of this Agreement pursuant to <u>Section 4(c)</u>.
- (b) <u>Partial Termination</u>. The Purchaser Party may terminate the provision of any Transition Service (in whole or in part) thirty (30) days following the date upon which the Purchaser Party notifies the Supplier Party in writing that the Purchaser Party no longer requires the Supplier Party or its Affiliates to provide such Transition Service or such shorter period of time as the Purchaser Party may determine with the consent of the Supplier Party, on behalf of itself and its Affiliates, in its sole discretion. Any election to terminate any Transition Service or a portion thereof shall not relieve the Purchaser Party of the obligation to pay for any terminated Transition Services that are provided prior to the effective date of such termination.
- (c) <u>Complete Termination</u>. The provision of Transition Services shall commence on the Effective Date and shall terminate upon the earliest to occur of the following (the "<u>Termination Date</u>"):
  - (i) the date that is six (6) months following the Effective Date;
- (ii) thirty (30) days following the date upon which the Purchaser Party notifies the Supplier Party in writing that the Purchaser Party no longer requires the Supplier Party to provide any Transition Services;
- (iii) fifteen (15) days following the date upon which the Supplier Party provides notice to the Purchaser Party that the Purchaser Party has failed to pay fees ("Failure to Pay") under this Agreement; and

- (iv) such other date as the Parties may mutually determine.
- (d) Effect of Termination or Expiration. Upon termination or expiration of this Agreement for any reason, the Supplier Party shall deliver to the Purchaser Party all records and other information pertaining to any matters for which the Supplier Party was providing Transition Services to the Purchaser Party hereunder; provided, however, that the Supplier Party may retain copies of such records and information to the extent necessary for accounting, tax reporting, compliance with the Supplier Party's document retention policies or other legitimate business purposes, subject to the requirements of Section 6 hereof. The Supplier Party acknowledges and agrees that after partial termination of this Agreement by the Purchaser Party with respect to any particular Transition Service, the Purchaser Party shall not have any payment obligations pursuant to Section 2 hereof with respect to such Transition Service performed after the effective date of such termination and that a partial termination of this Agreement by the Purchaser Party with respect to any particular Transition Service will not affect the Supplier Party's obligation to perform any other Transition Services hereunder.
- (e) <u>Survival</u>. <u>Section 4(d)</u> (Effect of Termination or Expiration), <u>Section 4(e)</u> (Survival), <u>Section 5</u> (Disclaimer of Warranties), <u>Section 6</u> (Indemnification), <u>Section 7</u> (Limitation of Liability), <u>Section 8</u> (Confidentiality) and <u>Section 10</u> (Miscellaneous) shall survive termination or expiration of this Agreement.

#### 5. Disclaimer of Warranties.

THIS IS A SERVICE AGREEMENT. EXCEPT AS EXPRESSLY STATED IN THIS AGREEMENT, (A) THE SUPPLIER PARTY DOES NOT GUARANTEE OR WARRANT THE TRANSITION SERVICES TO BE PROVIDED HEREUNDER, (B) THE TRANSITION SERVICES SHALL BE PROVIDED ON AN "AS IS" AND "WITH ALL FAULTS" BASIS AND (C) THERE ARE NO EXPRESS OR IMPLIED WARRANTIES OR GUARANTEES OF ANY KIND, INCLUDING ANY WARRANTY OF MERCHANTABILITY, NONINFRINGEMENT OR FITNESS FOR A PARTICULAR PURPOSE, AND ALL SUCH WARRANTIES ARE EXPRESSLY DISCLAIMED. THE SUPPLIER PARTY DOES NOT WARRANT THE CORRECTNESS. COMPLETENESS. GUARANTEE OR CURRENTNESS, MERCHANTABILITY, OR FITNESS FOR A PARTICULAR PURPOSE OF ANY DATA PROVIDED TO ANY OTHER PARTY.

#### 6. Indemnification.

The Purchaser Party hereby indemnifies the Supplier Party and each of its Affiliates against all damages, losses, liabilities and costs (including reasonable legal fees and expenses) attributable to any third party claims arising from or relating to the provision of the Transition Services, except to the extent that such damages, losses, liabilities or costs arise from the willful misconduct or gross negligence of the Supplier Party or any of its Affiliates, employees or subcontractors.

## 7. Limitation of Liability.

In no event shall the Supplier Party or any of its Affiliates, shareholders, directors, managers, officers, partners, members, employees, agents or representatives have any liability, whether based on warranty, contract, tort (including negligence or strict liability) or any other legal or equitable grounds, for any special, punitive, consequential, indirect, incidental or exemplary damages (including loss of data, profits, interest, revenue

or interruption of business) arising from or related to this Agreement or any of the Transition Services to be provided hereunder or the performance of or failure to perform the Supplier Party's obligations under this Agreement, regardless of whether such damages are foreseeable or an authorized representative of the Supplier Party is advised of the possibility or likelihood of such damages. In no event shall the aggregate liability of the Supplier Party or any of its Affiliates under this Agreement exceed the total amounts paid to the Supplier Party hereunder.

## 8. <u>Confidentiality</u>.

The Parties' obligations of confidentiality related to Transition Services shall be subject to and controlled by the Confidentiality Agreement.

### 9. <u>Force Majeure</u>.

If the Supplier Party is prevented from or delayed in complying, either totally or in part, with any of the terms or provisions of this Agreement by reason of fire, flood, storm, strike, lockout or other labor trouble, delays by unaffiliated suppliers or carriers, shortages of fuel, power, raw materials or components, any Applicable Law or Order. demand, seizure or requirement of any Governmental Authority, riot, civil commotion, war, rebellion, acts of terrorism, nuclear accident or other causes beyond the reasonable control of the Supplier Party, then (a) the Supplier Party shall advise the Purchaser Party as soon as practicable of the circumstances causing the disruption or delay, (b) the time for performing the obligations required by the affected terms or provisions of this Agreement shall be suspended during the period of such disability and (c) the Supplier Party shall have no liability to the Purchaser Party or any other party in connection therewith; provided that the Supplier Party shall take reasonable measures to overcome the disruption or delay with respect to the Purchaser Party which are consistent in all material respects with the measures taken in connection with the Supplier Party's business. In the event the Supplier Party is excused from supplying any Transition Service in accordance with the terms of this Section 9, the Purchaser Party shall be free to acquire such services from any substitute source at the Purchaser Party's expense. If the Supplier Party cannot perform any such delayed or disrupted Transition Service (or portion thereof) for a period of thirty (30) days due to such cause, then the Supplier Party may terminate this Agreement solely with respect to such delayed or disrupted Transition Service upon written notice to the Purchaser Party. If, after the period of disability, the Supplier Party is once again able to perform the Transition Services, the Supplier Party may provide notice to the Purchaser Party, and the Purchaser Party may reinstate this Agreement with respect to the Supplier Party's performance of such Transition Services upon written notice to the Supplier Party.

#### 10. Miscellaneous.

- (a) <u>Amendment</u>. No amendment of this Agreement shall be effective unless made in writing and signed by Seller and Purchaser.
- (b) <u>Waiver</u>. A waiver of any default, breach or non-compliance under this Agreement shall not be effective unless in writing and signed by the Party to be bound by the waiver, and then only in the specific instance and for the specific purpose for which it has been given. No waiver shall be inferred from or implied by any failure to act or delay in acting by a Party in respect of any default, breach or non-observance or by anything done or omitted to be done by the other Party. The waiver by a Party of any default, breach or non-compliance under this Agreement will not operate as a waiver of that Party's rights

under this Agreement in respect of any continuing or subsequent default, breach or non-observance (whether of the same or any other nature).

(c) <u>Notices</u>. All notices, demands and other communications to be given or delivered under or by reason of the provisions of this Agreement will be in writing and will be deemed to have been given (a) when personally delivered, (b) when transmitted via facsimile device or by electronic mail (unless if transmitted after 5:00 p.m. Central time or other than on a Business Day, then on the next Business Day), (c) the day following the day on which the same has been delivered prepaid to a reputable national overnight air courier service or (d) the third (3rd) Business Day following the day on which the same is sent by certified or registered mail, postage prepaid, in each case, to the respective party at the number, electronic mail address or street address, as applicable, set forth below, or at such other number, electronic mail address or street address as such party may specify by written notice to the other party hereto.

## Notices to the Supplier Party:

Seitel, Inc.

10811 South Westview Circle Drive

Bldg. C, #100

Houston, TX 77043

Attn: Marcia H. Kendrick

Email: [email address redacted – confidential information]

with a copy (which shall not constitute notice) to:

Kirkland & Ellis LLP 200 Clarendon Street Boston, MA 02116

Attention: Neal J. Reenan, P.C. and Amanda Border

Email: [email address redacted – confidential information]

[email address redacted – confidential information]

Kirkland & Ellis LLP 300 North LaSalle Street Chicago, IL 60654 Attention: Steve Toth

Email: [email address redacted – confidential information]

#### Notices to the Purchaser Party:

Seitel Canada Ltd.

c/o Pulse Seismic Inc.

Suite 2700, 421 – 7<sup>th</sup> Avenue S.W.

Calgary, AB T2P 4K9

Attention: Neal Coleman and Pamela Wicks

Email: [email address redacted – confidential information]

[email address redacted – confidential information]

with a copy (which shall not constitute notice) to:

Burgess Energy Advisors 200, 117 – 8<sup>th</sup> Avenue S.W.

Calgary, AB T2P 1B4

Attention: Patrick W. Burgess

Email: [email address redacted – confidential information]

- (d) <u>Successors and Assigns</u>. This Agreement shall enure to the benefit of, and be binding on, the Parties and their respective successors and permitted assigns. Neither Party may assign or transfer, whether absolutely, by way of security or otherwise, all or any part of its respective rights or obligations under this Agreement without the prior written consent of the other Party, other than to an Affiliate of such first Party.
- (e) <u>Severability</u>. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction will, as to that jurisdiction, be ineffective to the extent of such prohibition or unenforceability and will be severed from the balance of this Agreement, all without affecting the remaining provisions of this Agreement or affecting the validity or enforceability of such provision in any other jurisdiction.
- (f) <u>Third-Party Beneficiaries</u>. This Agreement is for the sole benefit of the Parties, and nothing in this Agreement, express or implied, is intended to or shall confer upon any other Person any legal or equitable right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- (g) <u>Entire Agreement</u>. This Agreement, the Purchase Agreement and the Confidentiality Agreement constitute the entire agreement between the Parties pertaining to the subject matter of this Agreement and supersedes all prior agreements, understandings, negotiations and discussions, whether oral or written. There are no conditions, representations, warranties, obligations or other agreements between the Parties in connection with the subject matter of this Agreement (whether oral or written, express or implied, statutory or otherwise).
- (h) <u>Counterparts</u>. This Agreement may be executed in counterparts, each of which shall be deemed to be an original and all of which taken together shall be deemed to constitute one and the same instrument. To evidence its execution of an original counterpart of this Agreement, a Party may send a copy of its original signature on the execution page hereof to the other Party by facsimile, e-mail in pdf format or by other electronic transmission and such transmission shall constitute delivery of an executed copy of this Agreement to the receiving Party.
- (i) Additional Rules of Interpretation. The inclusion in this Agreement of headings of sections are for convenience of reference only and are not intended to be full or precise descriptions of the text to which they refer. Wherever the words "include", "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation" and the words following "include", "includes" or "including" shall not be considered to set forth an exhaustive list. The words "hereof", "herein", "hereto", "hereunder", "hereby" and similar expressions shall be construed as referring to this Agreement in its entirety and not to any particular section or portion of it. All references herein to any agreement (including this Agreement), document or instrument mean such agreement, document or instrument as amended, supplemented, modified, varied, restated or replaced from time to time in accordance with the terms thereof and, unless otherwise specified therein, includes all schedules and exhibits attached thereto. The word "will" will be construed to have the same meaning and effect as the word "shall". The words "shall," "will," or "agree(s)" are mandatory, and "may" is permissive. The Parties hereto have participated jointly in the negotiation and drafting of this Agreement. In the event an

ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Parties, and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Agreement.

- Governing Law; Jurisdiction. All issues and questions concerning the construction, validity, interpretation and enforceability of this Agreement, and all claims and disputes arising hereunder or in connection herewith, whether purporting to sound in contract or tort, or at law or in equity, shall be governed by, and construed in accordance with, the laws of the State of Delaware, without giving effect to any choice of Applicable Law or conflict of Applicable Law rules or provisions (whether of the State of Delaware or any other jurisdiction) that would cause the application of the Applicable Laws of any iurisdiction other than the State of Delaware. The Parties hereby agree and consent to be subject to the exclusive jurisdiction of the Court of Chancery of the State of Delaware or, to the extent such court declines jurisdiction, first to any federal court, or second, to any state court, each located in Wilmington, Delaware, and hereby waive the right to assert the lack of personal or subject matter jurisdiction or improper venue in connection with any such suit, action or other proceeding. In furtherance of the foregoing, each of the Parties (a) waives the defense of inconvenient forum, (b) agrees not to commence any suit, action or other proceeding arising out of this Agreement or any transactions contemplated hereby other than in any such court, and (c) agrees that a final judgment in any such suit, action or other proceeding shall be conclusive and may be enforced in other jurisdictions by suit or judgment or in any other manner provided by Applicable Law.
- (k) <u>Waiver of Jury Trial</u>. EACH PARTY HEREBY IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY LITIGATION, ACTION, PROCEEDING, CROSS-CLAIM, OR COUNTERCLAIM IN ANY COURT (WHETHER BASED ON CONTRACT, TORT, OR OTHERWISE) ARISING OUT OF, RELATING TO OR IN CONNECTION WITH (I) THIS AGREEMENT OR THE VALIDITY, PERFORMANCE, INTERPRETATION, COLLECTION OR ENFORCEMENT HEREOF OR (II) THE ACTIONS OF SUCH PARTY IN THE NEGOTIATION, AUTHORIZATION, EXECUTION, DELIVERY, ADMINISTRATION, PERFORMANCE OR ENFORCEMENT HEREOF.
- (I) <u>Relationship of the Parties</u>. The Parties hereto are independent contractors and neither the Purchaser Party, on the one hand, nor the Supplier Party, on the other hand, is an employee, partner or joint venturer of the other. Under no circumstances shall any of the employees of the Purchaser Party or the Supplier Party be deemed to be employees of the other for any purpose. Neither the Purchaser Party nor the Supplier Party shall have the right to bind the other to any agreement with a third party nor to represent itself or themselves as a partner or joint venturer of the other.
- (m) <u>Purchase Agreement</u>. Nothing herein is intended to modify, limit or otherwise affect the representations, warranties, covenants, agreements and indemnifications contained in the Purchase Agreement, and such representations, warranties, covenants, agreements and indemnifications shall remain in full force and effect in accordance with the terms of the Purchase Agreement.

\* \* \* \* \*

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed by their duly authorized officers as of the date first set forth above.

SUPPLIER PARTY
SEITEL, INC.
By:
Name: Marcia H. Kendrick
Title: Chief Financial Officer, Executive Vice President,

Treasurer and Secretary

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed by their duly authorized officers as of the date first set forth above.

PURCHASER PARTY	
SEITEL CANADA LTD.	
By:	
Name:	

# **SCHEDULE A**

# TRANSITION SERVICES

Service	Cost	Time
Supplier Party will maintain the TSM Database, which contains the existing Seismic Data in archival format on tape, in Houston and in the same condition as maintained by Supplier Party immediately prior to Closing, and Purchaser Party may, upon request, receive a backup copy of the Seismic Data, in the standard form historically used by Supplier Party (i.e., tape format).	\$0.00	6 months
Purchaser Party may request the Seismic Data in the format required by Supplier Party and consistent with Supplier Party's standard historical practices, without customization.	Set forth on Schedule B	3 months
Supplier Party will forward emails that it receives that are sent to an individual that was formerly an employee of Seitel Canada, Ltd. post-Closing, excluding any such emails that contain Supplier Party's confidential and other sensitive information.	\$0.00	1 month
Supplier Party will provide Purchaser Party one (1) IT support person post-Closing solely to help with administrative transition matters related to local network, application passwords and account handover, and hardware, in accordance with a mutually agreed schedule of tasks to be performed.*	\$0.00	1 week

<sup>\*</sup> Supplier Party to offer extended services by IT personnel at a cost of \$2000 per day, if mutually agreed to extend the one week set forth in the table above. In no event shall such IT personnel be required to provide developing, migration or application oriented services, or any Excluded Services.

# **SCHEDULE B**

[Redacted – commercially sensitive information]

# Exhibit B

Escrow Agreement
[Attached]

#### **ESCROW AGREEMENT**

THIS AGREEMENT dated the 15th day of January, 2019

#### **BETWEEN**:

SEITEL CANADA HOLDINGS, INC.

(hereinafter "Vendor")

- and -

PULSE SEISMIC INC.

(hereinafter "Purchaser")

- and -

**BLAKE, CASSELS & GRAYDON LLP** 

(hereinafter "Escrow Agent")

#### WHEREAS:

- A. Vendor and Purchaser entered into a share purchase agreement as of January 15, 2019 (the "**Purchase Agreement**") regarding the sale of the Shares;
- B. The Escrow Agent has agreed to hold the Escrow Amount in escrow, to be released in accordance with the terms and conditions of this Agreement; and
- C. The Escrow Agent is willing to act as escrow agent for the sole purpose of dealing with the Escrow Amount in accordance with the terms and conditions of this Agreement.

**NOW THEREFORE** in consideration of the mutual covenants contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

- 1. **Definitions**. In this Agreement, including the recitals, unless otherwise provided herein, capitalized terms shall have the meanings given to them in the Purchase Agreement. In addition:
  - (a) "Transaction Parties" means Vendor and Purchaser and "Transaction Party" means any one of them, as the case may be.
- 2. **Appointment of Escrow Agent**. The Transaction Parties appoint Blake, Cassels & Graydon LLP to act as the Escrow Agent on the terms and conditions set forth in this Agreement and Blake, Cassels & Graydon LLP accepts such appointment on such terms and conditions.
- 3. **Receipt of Escrow Amount**. The Escrow Agent will notify the Transaction Parties once it confirms it is in receipt of the Escrow Amount and the Escrow Agent agrees to hold the

Escrow Amount in trust for the benefit of the Transaction Parties pursuant to this Agreement.

## 4. Disposition of the Escrow Amount.

- (a) The Escrow Agent is hereby irrevocably authorized and directed to disburse the Escrow Amount as set forth in this Section 4. The Escrow Agent will hold the Escrow Amount in accordance with the terms of this Agreement and will not distribute the Escrow Amount except in accordance with this Agreement.
- (b) The Transaction Parties shall deliver a joint written direction signed by both of the Transaction Parties (a "Joint Direction"), to the Escrow Agent and the Escrow Agent shall, within five (5) Business Days of receipt of a Joint Direction, release all or any portion of the Escrow Amount in accordance with such Joint Direction and any interest earned thereon shall be released to the Transaction Party(ies) receiving such payment (less any applicable withholding taxes) in the same proportion as the Escrow Amount to be released in connection therewith. Upon receipt by the Escrow Agent of a Joint Direction signed by both of the Transaction Parties, the Escrow Agent is hereby irrevocably authorized and directed to disburse all or any portion of the Escrow Amount in accordance with such Joint Direction. The Escrow Agent shall have no obligation to make any determination as to the validity of any such direction or any claim made by any Transaction Party for entitlement to the Escrow Amount and the Escrow Agent shall be entitled to continue to hold the Escrow Amount until such time as the Escrow Agent receives a Joint Direction signed by both the Transaction Parties as contemplated in this Section 4(b), until the Escrow Agent is directed by a Final Determination as to the disposition of the Escrow Amount, or until the Escrow Agent interpleads the Escrow Amount as contemplated in Section 6.
- 5. Interest. The Escrow Amount shall be held in escrow by the Escrow Agent and deposited in a daily interest bearing trust account with Canadian Imperial Bank of Commerce. The Escrow Agent makes no representation as to the yield available upon the Escrow Amount, shall bear no liability for any failure to achieve the maximum possible yield from the Escrow Amount and shall not be responsible for any failure of Canadian Imperial Bank of Commerce. Unless otherwise required by applicable law, Purchaser shall be treated as the owner of the Escrow Amount unless and until released pursuant to this Agreement, and, notwithstanding Section 4(b), all items of interest, if any, earned on the the Escrow Amount shall be taxable to Purchaser and Purchaser shall pay all income and other taxes applicable thereto or exigible thereon.
- 6. **Dispute**. In the event of any disagreement between the parties hereto which in the reasonable opinion of the Escrow Agent may result in adverse claims or demands with respect to the Escrow Amount or if any of the parties hereto are in disagreement about the interpretation of this Agreement or about the rights and obligations of the Escrow Agent or the proprietary of an action contemplated by the Escrow Agent under this Agreement, the Escrow Agent shall have the right at its option to deposit the Escrow Amount with the Accountant section of the Alberta Court of Queen's Bench (the "Court") in accordance with the Rules of Civil Procedure respecting interpleader or in such other manner or on such other grounds as such Court may direct. The Escrow Agent shall give written notice of any such deposit to the Transaction Parties immediately after such deposit is made.

- 7. **No Agency**. The Transaction Parties acknowledge that the Escrow Agent is acting solely as a stakeholder at their request and for their convenience and the Escrow Agent shall not be deemed to be the agent of any of the Transaction Parties in respect of the escrow herein referred to. The Escrow Agent shall not be liable to any of the Transaction Parties for any error in judgment or for any act or omission on its part in respect of the escrow herein referred to unless such error in judgment, act or omission is made, taken or suffered in bad faith or involves gross negligence, willful misconduct or fraud on the part of the Escrow Agent. For further clarity, in no event shall the actions of the Escrow Agent be seen as being taken in bad faith or involve gross negligence, wilful misconduct or fraud if the actions accord with directions in a Joint Direction.
- 8. **Indemnity**. Each Transaction Party hereby jointly and severally, agrees to indemnify and hold the Escrow Agent harmless from and against all costs, claims (including those from third parties) and expenses, including reasonable documented out-of-pocket solicitor's fees and disbursements incurred in connection with or arising from the performance of the Escrow Agent's duties or rights hereunder; provided that this indemnity shall not extend to actions or omissions taken or suffered by the Escrow Agent in bad faith or involving gross negligence, willful misconduct or fraud on the part of the Escrow Agent. Solely as between Vendor, on the one hand, and Purchaser, on the other hand, Vendor shall be responsible for fifty percent (50%) of such indemnification obligations and Purchaser shall be responsible for fifty percent (50%) of such indemnification obligations and each Transaction Party shall be entitled to reimbursement from the other to the extent the amount of Damages actually paid by any Transaction Party to an indemnified Transaction Party exceeds this agreed pro rata share.
- 9. **Fees**. Vendor and Purchaser each agree to pay to the Escrow Agent forthwith upon receipt of an invoice therefor fifty percent (50%) of \$[fee redacted]. In the event a Joint Direction is not executed and delivered to the Escrow Agent within sixty (60) days of the first anniversary of the date hereof, the Escrow Agent shall resign as Escrow Agent under Section 11 and if the Transaction Parties have not jointly appointed a new stakeholder in accordance with Section 11, the Escrow Agent shall deposit the Escrow Amount (including any interest earned thereon) with the Court in accordance with Section 6 less \$2,000 to cover costs associated with the Court process.
- 10. Limitation on Duties. It is understood and agreed that the Escrow Agent's only duties and obligations in respect of the Escrow Amount are expressly set out in this Agreement. The Escrow Agent shall have the right to consult with separate counsel of its own choosing (if it deems such consultation advisable) and shall not be liable for any action taken, suffered or omitted to be taken by it in good faith if the Escrow Agent acts in accordance with the advice of such counsel. The Escrow Agent shall be protected if it acts upon any written or oral communication, notice, certificate or other instrument or document believed by the Escrow Agent in good faith to be genuine and to be properly given or executed without the necessity of verifying the truth or accuracy of the same or the authority of the person giving or executing the same.
- 11. **Resignation of the Escrow Agent**. The Escrow Agent may, at any time, resign its obligations under this Agreement and be discharged from all further duties and liabilities hereunder by giving the Transaction Parties at least thirty (30) days' notice in writing of its intention to resign or such shorter notice as the Transaction Parties may accept as sufficient. The Transaction Parties agree that they shall as promptly as reasonably

practicable upon receipt of such notice jointly appoint a new stakeholder to act in the place and stead of the Escrow Agent and if they fail to agree on such appointment, any Transaction Party or the Escrow Agent may apply to a judge of the Court on such notice as such judge may direct for the appointment of a new stakeholder. Upon any new appointment, the new stakeholder will be vested with the same powers, rights, duties and obligations as if it had been originally named herein as stakeholder and such new stakeholder shall enter into an agreement with the Transaction Parties agreeing to be bound by all of the provisions of this Agreement. Upon any new appointment, the Escrow Agent shall deliver the Escrow Amount (including any interest earned thereon) to the new stakeholder.

- 12. **Termination.** Upon disposing of the Escrow Amount in accordance with the provisions of this Agreement, the Escrow Agent shall be relieved and discharged from all claims and liabilities relating to the Escrow Amount and the Escrow Agent shall not be subject to any claims made by or on behalf of any party hereto; provided, however, that the termination of this Agreement shall not relieve the Escrow Agent of its liabilities for any breach of this Agreement prior to such termination.
- 13. **No Conflict**. The fact that the Escrow Agent is acting as stakeholder under this Agreement shall not in any way prevent it from representing any of the Transaction Parties and/or their respective associates and affiliates in connection with the transactions, or other agreements, contemplated in the Purchase Agreement or any litigation arising from the Purchase Agreement, the other agreements contemplated thereby, or this Agreement or from representing each of the Transaction Parties and/or their respective associates and Affiliates or any other party in any other capacity or in any other transaction.
- 14. **Notice**. All notices or other communications given pursuant to this Agreement shall be in writing and shall be either delivered by hand or by facsimile transmission addressed as follows:
  - (a) in the case of Vendor, to:

Seitel Canada Holdings, Inc. 10811 South Westview Circle Drive Bldg. C, #100 Houston, TX 77043 Attn: Marcia H. Kendrick

Email: [email address redacted – confidential information]

with a copy (which shall not constitute notice) to:

Kirkland & Ellis LLP 200 Clarendon Street Boston, MA 02116

Attention: Neal J. Reenan, P.C. and Amanda Border

Email: [email address redacted – confidential information]

[email address redacted – confidential information]

Kirkland & Ellis LLP 300 North LaSalle Street Chicago, IL 60654 Attention: Steve Toth

Email: [email address redacted – confidential information]

(b) in the case of Purchaser, to:

Pulse Seismic Inc.

Suite 2700, 421 – 7<sup>th</sup> Avenue S.W.

Calgary, AB T2P 4K9

Attention: Neal Coleman and Pamela Wicks

Email: [email address redacted – confidential information]

[email address redacted – confidential information]

with a copy (which shall not constitute notice) to:

Burgess Energy Advisors 200, 117 – 8<sup>th</sup> Avenue S.W. Caldarv. AB T2P 1B4

Attention: Patrick W. Burgess

Email: [email address redacted – confidential information]

(c) in the case of the Escrow Agent, to:

Blake, Cassels & Graydon LLP 3500, 855 – 2<sup>nd</sup> Street SW Calgary, Alberta T2P 4J8

Attention: Ben Rogers and Noralee Bradley

Email: [email address redacted – confidential information]

[email address redacted – confidential information]

Any notice or other communication shall conclusively be deemed to have been given and received on the date on which it was delivered or sent if delivered or sent during normal business hours on a Business Day, and if delivered after normal business hours or on other than a Business Day, shall be deemed to have been given or sent on the next following Business Day. Any party may change its address for notices or other communications by giving notice thereof to the other parties to this Agreement in accordance with this Section 14. For the purposes hereof, "Business Day" means any day except Saturday, Sunday or a statutory holiday in Calgary, Alberta.

- 15. **Governing Law**. This Agreement shall be governed by and interpreted in accordance with the laws of the Province of Alberta and the federal laws of Canada applicable therein.
- 16. **Time**. Time shall be of the essence in this Agreement.
- 17. **Successors and Assigns**. This Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective successors and assigns.
- 18. **Amendment**. This Agreement shall only be amended by an instrument in writing executed and delivered by all parties.

- 19. **Waivers**. A waiver by a party at any time of its rights with respect to a default under this Agreement, or with respect to any other matter in connection with this Agreement, shall not be deemed a waiver with respect to any other or subsequent default or matter.
- 20. **Inurement**. This Agreement shall inure to the benefit of and be binding upon the parties and their respective successors (including any successor by reason of amalgamation of any party) and permitted assigns.
- 21. Further Assurances. The Parties shall with reasonable diligence and without additional consideration do all such things and provide all such reasonable assurances as may be required to consummate the transactions contemplated by this Agreement, and each party shall execute and deliver such further documents or instruments required by any other party as may be reasonably necessary or desirable to effect the purpose of this Agreement and carry out its provisions. Such further assurances may include, not more than once per calendar year, the provision of a commercially reasonable certificate to a prospective mortgagee or purchaser of an interest in this Agreement or a party relating to the effectiveness of this Agreement.
- 22. Counterparts. This Agreement may be executed and delivered in any number of counterparts with the same effect as if all signatories to the counterparts had signed one document, provided that no party is bound to this Agreement until all parties have executed and delivered a counterpart. All such counterparts shall together constitute and be construed as one instrument. For avoidance of doubt, a signed counterpart provided by way of facsimile transmission, e-mail transmission of Adobe Acrobat PDF files, or other electronic means shall constitute delivery of the counterpart once received and shall be as binding upon the parties as an originally signed counterpart.

[Signature Page Follows]

written	<b>TNESS WHEREOF</b> the parties have exect.	cuted this Agreement as o	of the date first above
SEITE	L CANADA HOLDINGS, INC.		
	Richard C. Kelvin President		

By:			
By: Name:			
Title:			

PULSE SEISMIC INC.

# **BLAKE, CASSELS & GRAYDON LLP**

By:		
Name:		
Title:		

# Exhibit C

Sample Calculation
[Attached]

	Estimated Working Capital at Close
Accounts Receivables	
A/R Other	[redacted]
GST	[redacted]
Refund	[redacted]
Prepaids and Other	[redacted]
Prepaid Expense	[redacted]
Prepaid Insurance	[redacted]
Deposits	[redacted]
	[redacted]
Total Estimated Current Assets	
Accounts Payable	
A/P Trade (invoiced)	[redacted]
A/P Trade (invoices not yet received)	[redacted]
GST Payable	[redacted]
	[redacted]
Employee Compensation Payable	
Commissions Payable	[redacted]
Bonus Payable	[redacted]
Vacation Accrual	[redacted]
Wages/payroll taxes payable and RRSP match	[redacted]
Income Taxes Receivable (Payable)	[redacted]
	[redacted]
Total Estimated Current Liabilities	[redacted]
Estimated Working Capital	[redacted]

[commercially sensitive information redacted]